POWER PURCHASE AGREEMENT

This Agreement ("Agreement" as further defined in Section 1.1) is made and entered into as of this ("Effective Date") and is witnessed and acknowledged by COMPANY with its principal office at ADDRESS__ ("Provider") and the State University of New York, an educational corporation organized and existing under the laws of the State of New York and having its principal place of business located at State University Plaza, Albany, New York, 12246 on behalf of SUNY – Purchase, located at 735 Anderson Hill Rd, Purchase, NY 10577 (the "Purchaser"), as evidenced by their signature on the last page of this document. Purchaser and Project Entity are referred to herein individually as a "Party" and collectively as the "Parties"

WITNESSETH:

WHEREAS, Purchaser desires that Provider install and operate a solar photovoltaic system and the ancillary equipment associated therewith at the Premises (as hereafter defined) ("System") for the purpose of providing Services (as hereafter defined), and Provider is willing to do the same;

WHEREAS Purchaser issued a Request for Proposal #SU052615 for the System, (the "RFP") which is attached hereto and made a part hereof as Exhibit C; and

WHEREAS, Provider has agreed to provide the System and submitted a Proposal dated DATE (the "Proposal") which is attached hereto and made a part hereof as Exhibit D. NOW THEREFORE, in consideration of the mutual promises set forth below, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

1. **DEFINITIONS.**

- 1.1 Definitions. In addition to other terms specifically defined elsewhere in the Agreement, where capitalized, the following words and phrases shall be defined as follows:
- "Actual Monthly Production" means the amount of Energy recorded by Provider's metering equipment during each calendar month of the Term, pursuant to Section 4.2.
- "Affiliate" means, with respect to any specified Person, any other Person directly or indirectly controlling, controlled by or under common control with such specified Person.
- "Agreement" means this Power Purchase Agreement, including the General Conditions and the Exhibits and Schedules attached hereto.
- "Applicable Law" means, with respect to any Person, any constitutional provision, law, statute, rule, regulation, ordinance, treaty, order, decree, judgment, decision, certificate, holding, injunction, registration, license, franchise, permit, authorization, guideline, Governmental Approval, consent or requirement of any Governmental Authority having jurisdiction over such Person or its property, enforceable at law or in equity, including the interpretation and administration thereof by such Governmental Authority.

"Assignment" has the meaning set forth in Section 13.1.

"Bankruptcy Event" means with respect to a Party, that either: (i) such Party has (A) applied for or consented to the appointment of, or the taking of possession by, a receiver, custodian, trustee or liquidator of itself or of all or a substantial part of its property; (B) admitted in writing its inability, or be generally unable, to pay its debts as such debts become due; (C) made a general assignment for the benefit of its creditors; (D) commenced a voluntary case under any bankruptcy law; (E) filed a petition seeking to take advantage of any other law relating to bankruptcy, insolvency, reorganization, winding up, or composition or readjustment of debts; (F) failed to controvert in a timely and appropriate manner, or acquiesced in writing to, any petition filed against such Party in an involuntary case under any bankruptcy law; or (G) taken any corporate or other action for the purpose of effecting any of the foregoing; or (ii) a proceeding or case has been commenced without the application or consent of such Party in any court of competent jurisdiction seeking (A) its liquidation, reorganization, dissolution or winding-up or the composition or readjustment of debts or, (B) the appointment of a trustee, receiver, custodian, liquidator or the like of such Party under any bankruptcy law, and such proceeding or case has continued undefended, or any order, judgment or decree approving or ordering any of the foregoing shall be entered and continue unstayed and in effect for a period of sixty (60) days.

"Business Day" means any day other than Saturday, Sunday or a Federal Reserve Bank holiday.

"Commercial Operation Date" has the meaning set forth in Section 3.3(b).

"Confidential Information" has the meaning set forth in Section 15.1.

"Covenants, Conditions and Restrictions" or "CCR" means those requirements or limitations related to the Premises as may be set forth in a lease, if applicable, or by any association or other organization, having the authority to impose restrictions.

"Disruption Period" has the meaning set forth in Section 4.3(b).

"Early Termination Date" means any date on which the Agreement terminates other than by reason of expiration of the then applicable Term.

"Early Termination Fee" means the fee payable by Purchaser to Provider under the circumstances described in Section 2.2, Section 4.3(a) or Section 11.2.

"Energy" means electric energy measured in kilowatt-hours ("kWh") or in megawatt-hours ("MWh").

"Environmental Attributes" means, an aspect, claim, characteristic or benefit associated with the generation of a quantity of electricity by an electricity generation facility that is capable of being measured, verified or calculated, including any and all credits, benefits, emissions reductions, offsets and allowances, howsoever entitled, attributable to the generation of such quantity of electricity by an electricity generation facility and its displacement of conventional, non-renewable electricity generation together with the right(s) to report ownership of such attributes to any agency, authority, or third party. Environmental Attributes shall not include (i) any energy, capacity, reliability or other power attributes from the electricity generation facility; (ii)

production tax credits associated with the construction or operation of the electricity generation facility and other financial incentives in the form of credits, reductions or allowances associated with the electricity generation facility that are applicable to a state, provincial or federal income taxation obligation; or (iii) fuel-related subsidies, "tipping fees", or other local subsidies received by the electricity generation facility for the destruction of particular preexisting pollutants or the promotion of local environmental benefits

"Estimated Annual Production" has the meaning set forth in Section 5.2.

"Expiration Date" means the date on which the Agreement terminates by reason of expiration of the Term.

"Fair Market Value" means, with respect to any tangible asset or service, the price that would be negotiated in an arm's-length, free market transaction, for cash, between an informed, willing seller and an informed, willing buyer, neither of whom is under compulsion to complete the transaction. Fair Market Value of the System will be determined pursuant to Section 2.4.

"Financing Party" means, as applicable (i) any Person (or its agent) from whom Provider (or an Affiliate of Provider) leases the System, or (ii) any Person (or its agent) who has made or will make a loan to or otherwise provider financing to Provider (or an Affiliate of Provider) with respect to the System.

"Force Majeure Event" has the meaning set forth in Section 10.1.

"General Conditions" means this Power Purchase Agreement, excluding the Exhibits and Schedules hereto.

"Governmental Approval" means any approval, consent, franchise, permit, certificate, resolution, concession, license, or authorization issued by or on behalf of any applicable Governmental Authority.

"Governmental Authority" means any federal, state, regional, county, town, city, or municipal government, whether domestic or foreign, or any department, agency, bureau, or other administrative, regulatory or judicial body of any such government.

"Indemnified Persons" means the Purchaser Indemnified Parties or the Provider Indemnified Parties, as the context requires.

"Initial Term" has the meaning set forth in Section 2.1.

"Installation Work" means the construction and installation of the System and the start-up, testing and acceptance (but not the operation and maintenance) thereof, all performed by or for Provider at the Premises.

"Invoice Date" has the meaning set forth in Section 6.2.

"kWh Rate" means the price per kWh set forth in Schedule 2 payable for the services to be provided hereunder including Energy, and heat.

"Liens" has the meaning set forth in Section 7.1(e).

"Letter(s) of Credit" means one or more irrevocable, non-transferable standby letters of credit issued by a U.S. commercial bank or a U.S. branch of a foreign bank (which is not an Affiliate of either Party) with such bank having a credit rating of at least A- from Standard & Poor's Rating Group and A3 from Moody's Investor Service, and otherwise being in a form acceptable to the Party in whose favor the letter of credit is issued. Costs of a Letter of Credit shall be borne by the applicant for such Letter of Credit.

"Local Electric Utility" means the local electric distribution owner and operator providing electric distribution and interconnection services to Purchaser at the Premises.

"Losses" means all losses, liabilities, claims, demands, suits, causes of action, judgments, awards, damages, cleanup and remedial obligations, interest, fines, fees, penalties, costs and expenses (including all reasonable attorneys' fees and other reasonable costs and expenses incurred in defending any such claims or other matters or in asserting or enforcing any indemnity obligation).

"Option Price" has the meaning set forth in Section 2.3.

"Party" or "Parties" has the meaning set forth in the preamble hereof.

"Person" means an individual, partnership, corporation, limited liability company, business trust, joint stock company, trust, unincorporated association, joint venture, firm, or other entity, or a Governmental Authority.

"Premises" means the premises described in Schedule 1 of the Special Conditions. For the avoidance of doubt, the Premises includes the entirety of any structures and underlying real property located at the address described in Schedule 1.

"Provider" has the meaning set forth in the Special Conditions.

"Provider Default" has the meaning set forth in Section 11.1(a).

"Provider Indemnified Parties" has the meaning set forth in Section 16.2.

"Purchase Date" means the day that occurs on the date that is the sixth (6th) anniversary, the tenth (10th) anniversary, and the fifteenth (15th) anniversary of the Commercial Operation Date, provided that if such day is not a Business Day, the Purchase Date shall be the next Business Day to occur after such day.

"Purchaser" has the meaning set forth in the Special Conditions.

"Purchaser Default" has the meaning set forth in Section 11.2(a).

"Purchaser Indemnified Parties" has the meaning set forth in Section 16.1.

"Renewal Term" has the meaning set forth in Section 2.1.

- "Representative" has the meaning set forth in Section 15.1.
- "Security Agreement" has the meaning set forth in Section 8.2.
- "Stated Rate" means a rate per annum equal to the lesser of (a) the "prime rate" (as reported in The Wall Street Journal) plus two percent (2%) and (b) the maximum rate allowed by Applicable Law.
- "System Operations" means the Provider's operation, maintenance and repair of the System performed in accordance the requirements herein.
- "Term" has the meaning set forth in Section 2.1.
- "Transfer Time" has the meaning set forth in Section 4.3(a).
- 1.2 Interpretation. The captions or headings in these General Conditions are strictly for convenience and shall not be considered in interpreting the Agreement. Words in the Agreement that impart the singular connotation shall be interpreted as plural, and words that impart the plural connotation shall be interpreted as singular, as the identity of the parties or objects referred to may require. The words "include", "includes", and "including" mean include, includes, and including "without limitation" and "without limitation by specification." The words "hereof", "herein", and "hereunder" and words of similar import refer to the Agreement as a whole and not to any particular provision of the Agreement. Except as the context otherwise indicates, all references to "Articles" and "Sections" refer to Articles and Sections of these General Conditions.
- 1.3 Entire Agreement and Order of Precedence. This Agreement forms the entire agreement between the parties and supersedes all written or oral, prior or contemporaneous communications between the parties relating to the subject matter of this Agreement.
- In case of a conflict or discrepancy among the elements of this Agreement, such conflict or inconsistency shall be resolved by giving precedence to the document elements in the the following order: (1) Exhibits A and A-1; (2) Exhibit B, (3) the Schedules, (4) the General Conditions, (5) the RFP, and (6) the Proposal.

2. TERM AND TERMINATION.

- 2.1 Term. The term of the Agreement shall commence on the Effective Date and shall continue for twenty-five (25) years from the Commercial Operations Date ("Initial Term"), unless and until terminated earlier pursuant to the provisions of the Agreement. After the Initial Term, the Agreement shall be-renewed by the parties for an additional five (5) year term (a "Renewal Term"). The Initial Term and the subsequent Renewal Term, if any, are referred to collectively as the "Term." During any Renewal Term, either Party may, subject to Section 2.2, terminate the Agreement upon one hundred and eighty (180) days' prior written notice to the other Party.
- 2.2 Early Termination. Purchaser may terminate the Agreement prior to any applicable Expiration Date for any reason upon sixty (60) days' prior written notice. In such event, Purchaser shall pay, as liquidated damages, an amount equal to (i) the Early Termination Fee set

forth on Schedule 3, (ii) any and all other amounts previously owed to Provider under this Agreement, and (iii) the cost of removing the System. Upon such payment, Provider shall cause the System to be disconnected and removed from the Premises. Upon Purchaser's payment to Provider of the Early Termination Fee, the Agreement shall terminate automatically without further liability to either Party with respect to the Agreement.

- 2.3 Purchase Option. On any Purchase Date, so long as a Purchaser Default shall not have occurred and be continuing, Purchaser has the option to purchase the System for a purchase price (the "Option Price") equal to the greater of (a) the Fair Market Value of the System as of the Purchase Date and (b) the purchase price applicable for such Purchase Date as set forth in Schedule 4, Column 2i. To exercise its purchase option, Purchaser shall, not less than one hundred and eighty (180) days prior to the proposed Purchase Date, provide written notice to Provider of Purchaser's intent to exercise its option to purchase the System on such Purchase Date. Within thirty (30) days of receipt of Purchaser's notice, Provider shall specify the Option Price as determined by Provider in a commercially reasonable manner, and Purchaser shall then have a period of thirty (30) days after notification of the Option Price to confirm or retract its decision to exercise the purchase option or, if, in Purchaser's sole opinion, the Option Price stated by Provider is not equal to the Fair Market Value of the System, to dispute the determination of the Fair Market Value of the System. In the event Purchaser confirms its exercise of the purchase option in writing to Provider (whether before or after any determination of the Fair Market Value determined pursuant to Section 2.4), (i) the Parties shall promptly execute all documents necessary to (A) cause title to the System to pass to Purchaser on the Purchase Date, free and clear of any Liens, and (B) assign all vendor warranties for the System to Purchaser, and (ii) Purchaser shall pay the Option Price to Provider on the Purchase Date, such payment to be made in accordance with any previous written instructions delivered to Purchaser by Provider or Provider's Financing Party, as applicable, for payments under the Agreement. Upon execution of the documents and payment of the Option Price, in each case as described in the preceding sentence, the Agreement shall terminate automatically without further liability to either Party with respect to the Agreement. For the avoidance of doubt, payment of the Option Price shall be in lieu of and instead of any payments as described in Section 2.2 hereof. In the event Purchaser retracts its exercise of, or does not timely confirm, the purchase option, the provisions of the Agreement shall remain applicable as if the Purchaser had not exercised any option to purchase the System.
- 2.4 Determination of Fair Market Value. If the Option Price indicated by Provider in accordance with Section 2.3 is equal to the Fair Market Value (as determined by Provider in a commercially reasonable manner) and Purchaser disputes such stated Fair Market Value within thirty (30) days of receipt of such notice from Provider, then the Parties shall mutually select an independent appraiser with experience and expertise in the commercial-scale photovoltaic industry. Such appraiser shall subject to applicable Internal Revenue Service processes and requirements act_in a commercially reasonable manner and in good faith to determine the Fair Market Value of the System as of the applicable Purchase Date and shall set forth such determination in a written opinion delivered to the Parties. The valuation made by the appraiser shall be binding upon the Parties in the absence of fraud or manifest error. The costs of the appraisal shall be borne by Purchaser if such appraisal results in a value equal or greater than the value provided by Provider pursuant to Section 2.3; otherwise, the Parties shall equally share such cost.

- 2.5 Removal of System at Expiration. Subject to Purchaser's exercise of its purchase option under Section 2.3, upon the expiration or earlier termination of the Agreement, Provider shall, at Provider's expense, remove all of its tangible property comprising the System from the Premises on a mutually convenient date but in no case later than one hundred and twenty (120) days after the Expiration Date. The Premises shall be returned to its original condition, except for System mounting pads or other support structures and ordinary wear and tear. If the System is to be located on a roof, then in no case shall Provider's removal of the System affect the integrity of Purchaser's roof, which shall be as leak-proof and structurally sound as it was prior to removal of the System (other than ordinary wear and tear). For purposes of Provider's removal of the System, Purchaser's covenants pursuant to Section 7.2 shall remain in effect until the date of actual removal of the System. Provider shall leave the Premises in neat and clean order. If Provider fails to remove or commence substantial efforts to remove the System by such agreed upon date, Purchaser shall have the right, at its option, to remove the System mounting pads or other support structures and ordinary wear and tear) at Provider's cost.
- 2.6 Conditions of the Agreement Prior to Installation. In the event that any of the following events or circumstances occur prior to the Commercial Operation Date, Provider may (at its sole discretion) terminate the Agreement, in which case neither Party shall have any liability to the other except for any such liabilities that may have accrued prior to such termination: (a) the Provider determines that the Premises, as is, is insufficient to accommodate the System; (b) there exist site conditions (including environmental conditions) or construction requirements that were not known as of the Effective Date and that could reasonably be expected to materially increase the cost of Installation Work or would adversely affect the electricity production from the System as designed; (c) there is a material adverse change in the regulatory environment, incentive program or federal or state tax code (including the expiration of any incentive program or tax incentives in effect as of the Effective Date) that could reasonably be expected to adversely affect the economics of the installation for Provider and its investors; (d) Provider is unable to obtain financing for the System on terms and conditions satisfactory to it; (e) Provider has not received a fully executed (i) license substantially in the form of Exhibit B from the owner of the Premises, and (ii) a release or acknowledgement from any mortgagee of the Premise, if required by Provider's Financing Party, to establish the priority of its security interest in the System, and (iii) such other documentation as may be reasonably requested by Provider to evidence Purchaser's ability to meet its obligations under Section 7.2(d)(ii) to ensure that Provider will have access to the Premises throughout the Term; (f) there has been a material adverse change in the rights of Purchaser to occupy the Premises or Provider to construct the System on the Premises; (g) Purchaser has not received evidence reasonably satisfactory to it that interconnection services will be available with respect to Energy generated by the System under a net-metering arrangement mutually agreeable to the Parties, subject to applicable law and the utility's requirements; or (h) Purchaser has determined that there are easements, CCRs or other liens or encumbrances that would materially impair or prevent the installation, operation, maintenance or removal of the System.

3. PURCHASER RIGHT TO REVIEW CONSTRUCTION, INSTALLATION AND TESTING OF SYSTEM.

- 3.1 Installation Work. Provider will cause the System to be designed, engineered, installed and constructed substantially in accordance with Schedule 1, Applicable Law, and consistent with generally accepted utility practices. Upon request, Purchaser shall have the right to review all construction plans and designs, including engineering evaluations of the impact of the System. Provider shall perform the Installation Work at the Premises between the hours of 7:00 a.m. and 7:00 p.m. unless otherwise agreed to in writing by the owner of the Premises. Installation Work shall be conducted in a manner that minimizes inconvenience to and interference with the use of the Premises to the maximum extent commercially practical.
- 3.2 Approvals; Permits. Purchaser shall reasonably assist Provider in obtaining all necessary approvals and permits including but not limited to those related to the Local Electric Utility, any Governmental Authority, and any waivers, approvals or releases required pursuant to any applicable CCR.
- 3.3 System Acceptance Testing. (a) Provider shall conduct testing of the System in accordance with such methods, acts, guidelines, standards and criteria reasonably accepted or followed by commercial solar photovoltaic energy system integrators and operators in the United States. (b) If the results of such testing indicate that the System is capable of generating Energy for at least four (4) consecutive hours, using such instruments and meters as have been installed for such purposes, and the System has been approved for interconnected operation by the Local Electric Utility and any other Governmental Authority from whom approval is required to operate the System, then Provider shall send a written notice to Purchaser to that effect, and the date of such notice shall be the "Commercial Operation Date."

4. SYSTEM OPERATIONS.

- 4.1 Provider as Owner and Operator. The System will be owned by Provider or Provider's Financing Party and will be operated and maintained and, as necessary, repaired by Provider at its sole cost and expense; provided, that any repair or maintenance costs incurred by Provider as a result Purchaser's negligence or breach of its obligations hereunder shall be reimbursed by Purchaser.
- 4.2 Metering. Provider shall install and maintain a utility grade kilowatt-hour (kWh) meter for the measurement of electrical energy provided by the System.
- 4.3 System Disruptions. (a) Substitution of Premises. If, for reasons other that Provider's negligence, willful misconduct, or breach of its obligations hereunder or as otherwise provided for in this Agreement, Provider ceases to have access rights to the Premises as necessary to operate the System prior to the Expiration Date, then Purchaser shall either (i) provide Provider with a mutually agreeable substitute premises in a location with similar capability to accommodate the System, or (ii) terminate the Agreement pursuant to Section 2.2. In connection with such substitution, Purchaser and Provider shall amend this Agreement to specify the substitute premises. Purchaser shall also provide any new owner, lessor, or mortgagee consents or releases required by Provider's Financing Party in connection with the substitute premises. If Purchaser is unable to obtain such consents and releases for a substitute premises, the substitution shall not be allowed and Purchaser shall terminate the Agreement pursuant to Section 2.2. Purchaser shall pay all reasonable costs associated with relocation of the System

including all costs and expenses incurred by or on behalf of Provider in connection with removal of the System from the existing Premises and repair or maintenance of the Premises, if applicable, and installation and testing of the System at such substitute premises and all applicable interconnection fees and expenses at the substitute premises, as well as costs of new title search and other out of pocket expenses connected to preserving and re-filing the security interest of Provider's Financing Party in the System. Provider shall make commercially reasonable efforts to remove all of its tangible property comprising the System from the vacated Premises prior to the termination of Purchaser's rights to use such Premises. Upon removal of the tangible property comprising the System from the Premises, at the Purchaser's expense, the Premises shall be returned to its original condition, except for incidental hardware or other support structures and ordinary wear and tear. In connection with any substitution of Premises, Purchaser shall continue to make all payments for the Services as if Provider was capable of operating the System during any transfer or construction time period (the "Transfer Time") and shall reimburse Provider for any actual, documented lost or recaptured revenue associated with lost production of Environmental Attributes during the Transfer Time. For the purpose of calculating Services Payments that would have otherwise been due and Environmental Attributes that would have been generated during such Transfer Time, Services shall be deemed to have been produced during the Transfer Time at the average rate over the preceding twelve (12) months (or, if the substitution occurs within the first twelve (12) months of operation, the average over such period of operation), discounting for any planned or other maintenance outages of the System that had been scheduled by the Provider to occur during such Transfer Time.

(b) System Disruptions. In the event that (1) the owner or lessee of the Premises repairs the Premises for any reason not directly related to damage caused by the System, and such repair requires the partial or complete temporary disassembly or movement of the System, or (2) any act or omission of Purchaser or Purchaser's employees, Affiliates, or agents (collectively, a "Purchaser Act") results in a disruption or outage in System production, then, in either case, Purchaser shall (i) pay Provider for all work required by Provider to disassemble or move the System and (ii) continue to make all payments for the Services during such period of System disruption (the "Disruption Period") as if Provider was capable of operating the System during such Disruption Period, and (iii) reimburse Provider for any actual, documented lost or recaptured revenue associated with lost production of Environmental Attributes during the Disruption Period. For the purpose of calculating Services Payments that would otherwise have been due for such Disruption Period, Services shall be deemed to have been produced at the average rate over the preceding twelve (12) months (or, if the disruption occurs within the first twelve (12) months of operation, the average over such period of operation), discounting for any planned or other maintenance outages of the System that had been scheduled by the Provider to occur during such Disruption Period.

5. DELIVERY OF SERVICES.

5.1 Purchase Requirement. Purchaser agrees to purchase one hundred percent (100 %) of the Energy generated by the System during each relevant month of the Term at the kWh Rate specified in Schedule 2. The Energy delivered by Provider to Purchaser hereunder shall be referred to as "Services".

- 5.2 Estimated Annual Production. The annual estimate of Energy generation with respect to the System for any given year as determined pursuant to this Section shall be the "Estimated Annual Production." The Estimated Annual Production for each year of the Term is set as forth in Schedule 4.
- 5.3 Environmental Attributes. Purchaser's purchase of Services does not include Environmental Attributes associated with the Energy generated by the System, which shall be retained by Provider or Provider's Financing Party for the duration of the Term. Purchaser disclaims any right to such Environmental Attributes based upon the installation of the System at the Premises, and shall, at the request of Provider, execute any document or agreement reasonably necessary to fulfill the intent of this Section 5.3. To avoid any conflicts with fair trade rules regarding claims of renewable energy use and to help ensure that Environmental Attributes will be certified by Green-e® or a similar organization, Purchaser shall submit to Provider for approval any press releases regarding Purchaser's use of renewable energy from the System and shall not submit for publication any such releases without the prior written approval of Provider. 5.4 Title to System. Subject to Purchaser's exercise of its purchase option under Section 2.3, throughout the Term, Provider or Provider's Financing Party shall be the legal and beneficial owner of the System at all times, and the System shall remain the personal property of Provider or Provider's Financing Party and shall not attach to or be deemed a part of, or fixture to, the Premises. The System shall at all times retain the legal status of personal property as defined under Article 9 of the Uniform Commercial Code. Purchaser covenants that it will use reasonable commercial efforts to place all parties having an interest in or lien upon the real property comprising the Premises on notice of the ownership of the System and the legal status or classification of the System as personal property. If there is any mortgage or fixture filing against the Premises which could reasonably be construed as attaching to the System as a fixture of the Premises, Purchaser shall provide, at Provider's request, a disclaimer or release from such lien holder. If Purchaser is the fee owner of the Premises, Purchaser agrees to file, at the request of Provider and at Provider' sole expense, a disclaimer of the System as a fixture of the Premises in the office where real estate records are customarily filed in the jurisdiction of the Premises. If Purchaser is not the fee owner, Purchaser will, at Provider's request, use commercially reasonable efforts to obtain such agreement to file from such owner or consent of such owner to have Provider make such filing on its behalf; provided that such filing will be made at Provider's request and at Provider's sole expense.

6. PRICE AND PAYMENT.

6.1 Consideration. Purchaser shall pay to Provider a monthly payment for the Services provided during each month in an amount equal to the Energy generated by the System and delivered to Purchaser in such month, multiplied by the kWh Rate stated in Schedule 2 for the applicable contract year in which such month occurs ("Services Payment").

In the event that Purchaser is a municipality or other Governmental Authority, if sufficient funds to provide for payment(s) owed by Purchaser under this Agreement are not appropriated, the Purchaser shall terminate this Agreement upon notice in writing to Provider in accordance with the terms of Section 2.2, including, without limitation, the payment to Provider of the Early Termination Fee.

- 6.2 Invoice. Provider shall invoice Purchaser on or about the first day of each month (each, an "Invoice Date"), commencing on the first Invoice Date to occur after the Commercial Operation Date, for the Services Payment in respect of the immediately preceding month. The last invoice shall include production only through the Expiration Date of this Agreement.
- 6.3 Time of Payment. Purchaser shall pay all undisputed amounts due hereunder within thirty (30) days after the date of the applicable Invoice Date.
- 6.4 Method of Payment. Purchaser shall make all payments under the Agreement by electronic funds transfer in immediately available funds to the account designated by Provider from time to time. Interest for any unpaid balance will accrue pursuant to Section 179 (G) of New York State Finance Law that is 30 days past due. All payments made hereunder shall be non-refundable, be made free and clear of any tax, levy, assessment, duties or other charges and not subject to reduction, withholding, set-off, or adjustment of any kind.
- 6.5 Disputed Payments. If a bona fide dispute arises with respect to any invoice, Purchaser shall not be deemed in default under the Agreement and the Parties shall not suspend the performance of their respective obligations hereunder, provided that Purchaser shall not be required to pay the disputed amounts owed hereunder until such dispute has been resolved by the Parties. If an amount disputed by Purchaser is subsequently deemed to have been due pursuant to the applicable invoice, interest shall accrue on such amount pursuant to Section 179 (G) of New York State Finance Law from the date becoming past due under such invoice until the date paid

7. GENERAL COVENANTS.

- 7.1 Provider's Covenants. Provider covenants and agrees to the following: (a) Notice of Damage or Emergency. Provider shall (i) promptly notify Purchaser if it becomes aware of any damage to or loss of the use of the System or that could reasonably be expected to adversely affect the System, (ii) immediately notify Purchaser it becomes aware of any event or circumstance that poses an imminent risk to human health, the environment, the System or the Premises in connection with this Agreement.
- (b) System Condition. Provider shall take all actions reasonably necessary to ensure that the System is capable of providing Services at a commercially reasonable continuous rate.
- (c) Governmental Approvals. While providing the Installation Work, Services, and System Operations, Provider shall obtain and maintain and secure all Governmental Approvals required to be obtained and maintained and secured by Provider and to enable Provider to perform such obligations.
- (d) Health and Safety. Provider shall take all necessary and reasonable safety precautions with respect to providing the Installation Work, Services, and System Operations and shall comply with all Applicable Laws pertaining to the health and safety of persons and real and personal property.
- (e) Liens. Other than a Financing Party's security interest in or ownership of the System, Provider shall not directly or indirectly cause, create, incur, assume or suffer to exist any mortgage, pledge, lien (including mechanics', labor or materialman's lien), charge, security interest, encumbrance or claim of any nature ("Liens") on or with respect to the Premises or any interest therein, in each case to the extent such Lien arises from or is related to Provider's performance or non-performance of its obligations hereunder. If Provider breaches its

obligations under this Section, it shall (i) immediately notify Purchaser in writing, (ii) promptly cause such Lien to be discharged and released of record without cost to Purchaser, and (iii) defend and indemnify Purchaser against all costs and expenses (including reasonable attorneys' fees and court costs at trial and on appeal) incurred in discharging and releasing such Lien.

- 7.2 Purchaser's Covenants. Purchaser covenants and agrees as follows: (a) Notice of Damage or Emergency. Purchaser shall (i) promptly notify Provider if it becomes aware of any damage to or loss of the use of the System or that could reasonably be expected to adversely affect the System; (ii) immediately notify Provider if it becomes aware of any event or circumstance that poses an imminent risk to human health, the environment, the System or the Premises in connection with this Agreement.
- (b) Liens. Other than as provided for in this Agreement, Purchaser shall not directly or indirectly cause, create, incur, assume or suffer to exist any Liens on or with respect to the System or any interest therein. If Purchaser breaches its obligations under this Section, it shall immediately notify Provider in writing, shall promptly cause such Lien to be discharged and released of record without cost to Provider, and subject to the availability of lawful appropriations and consistent with Section 8 of the State Court of Claims Act, Purchaser shall hold Provider harmless from and indemnify it for any final judgment of a court of competent jurisdiction to the extent attributable to the negligence of the Purchaser or of its officers or employees when acting within the course and scope of their employment.
- (c) Consents and Approvals. Purchaser or Provider shall ensure that any authorizations required of Purchaser or Provider under this Agreement are provided in a timely manner. To the extent that only Purchaser is authorized to request, obtain or issue any necessary approvals, permits, rebates or other financial incentives, Purchaser shall reasonably cooperate with Provider, at Provider's request, to obtain such approvals, permits, rebates or other financial incentives.
- (d) Access to Premises, Grant of License. Purchaser hereby grants to Provider a commercial license (the "License") commencing on the Effective Date and, subject to Purchaser's right to revoke the License as described herein, continuing until the date that is one hundred and twenty (120) days after the expiration or termination of this Agreement (the "License Term"), containing all the rights necessary for Provider to use and occupy portions of the Premises (the "License Area") for the installation, operation and maintenance of the System pursuant to the terms and conditions of this Agreement, including ingress and egress rights to the Premises for Provider and its employees, and access to electrical panels and conduits to interconnect or disconnect the System with the Premises' electrical wiring. During the License Term Purchaser shall ensure that Provider's rights under the License and Provider's access to the Premises are preserved and protected and shall not interfere with or permit any third parties to interfere with such rights or access. Regardless of whether Purchaser is owner of the Premises or leases the Premises from a landlord, Purchaser hereby covenants that (i) Provider shall have access to the Premises and System during the Term of this Agreement and for so long as needed after termination to remove the System pursuant to the applicable provisions herein, and (ii) neither Purchaser nor Purchaser's landlord will interfere or handle any Provider equipment or the System without written authorization from Provider; provided, however, that Purchaser and Purchaser's landlord shall at all times have access to and the right to observe the Installation Work or System removal. Notwithstanding anything to the contrary in this Agreement, Purchaser may, at any time during the Term, upon 60 days' prior written notice to Provider, revoke the License, provided that any such revocation of the License, except for a revocation for a Provider Default, shall act as an early termination of this Agreement pursuant to Section 2.2.

- (e) Temporary storage space during installation or removal. Purchaser shall use commercially reasonable efforts to obtain an agreement for sufficient space at the Premises for the temporary storage and staging of tools, materials and equipment and for the parking of construction crew vehicles and temporary construction trailers and facilities reasonably necessary during the Installation Work, System Operations or System removal, and access for rigging and material handling.
- (f) Insolation. Purchaser understands that unobstructed access to sunlight ("<u>Insolation</u>") is essential to Provider's performance of its obligations and a material term of this Agreement. Purchaser shall not in any way cause and, where possible, shall not in any way permit any interference with the System's Insolation. If Purchaser becomes aware of any activity or condition that could diminish the Insolation of the System, Purchaser shall notify Provider immediately and shall cooperate with Provider in preserving the System's existing Insolation levels.

8. REPRESENTATIONS & WARRANTIES.

- 8.1 Representations and Warranties Relating to Agreement Validity. In addition to any other representations and warranties contained in the Agreement, each Party represents and warrants to the other as of the Effective Date that:
- (a) it is duly organized and validly existing and in good standing in the jurisdiction of its organization;
- (b) it has the full right and authority to enter into, execute, deliver, and perform its obligations under the Agreement and shall maintain such right and authority throughout the Term;
- (c) it has taken all requisite corporate or other action to approve the execution, delivery, and performance of the Agreement;
- (d) the Agreement constitutes its legal, valid and binding obligation enforceable against such Party in accordance with its terms, except as may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, and other similar laws now or hereafter in effect relating to creditors' rights generally;
- (e) there is no litigation, action, proceeding or investigation pending or, to the best of its knowledge, threatened before any court or other Governmental Authority by, against, affecting or involving any of its business or assets that could reasonably be expected to materially adversely affect its ability to perform its obligations hereunder or carry out the transactions contemplated herein; and
- (f) its execution and performance of the Agreement and the transactions contemplated hereby do not constitute a breach of any term or provision of, or a default under, (i) any contract or agreement to which it or, in the case of Provider, any of its Affiliates is a party or by which it or, in the case of Provider, any of its Affiliates or its or their property is bound, (ii) its organizational documents, or (iii) any Applicable Laws.
- 8.2 Representations Regarding Security Interest. Purchaser has been advised that part of the collateral securing the financial arrangements for the System may be the granting of a first priority perfected security interest (the "Security Interest") in the System to a Financing Party. In connection therewith, Purchaser represents and warrants as follows:

- (a) to Purchaser's knowledge, the granting of the Security Interest will not violate any term or condition of any covenant, restriction, lien, financing agreement, or security agreement affecting the Premises:
- (b) Purchaser is aware of no existing lease, mortgage, security interest or other interest in or lien upon the Premises that could attach to the System as an interest adverse to Provider's Financing Party's Security Interest therein;
- (c) to Purchaser's knowledge, there exists no event or condition which constitutes a default, or would, with the giving of notice or lapse of time, constitute a default under this Agreement. Any Financing Party shall be an intended third party of this Section 8.2.
- 8.3 EXCEPT AS EXPRESSLY SET FORTH HEREIN, THERE IS NO WARRANTY OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, AND ANY AND ALL IMPLIED WARRANTIES ARE DISCLAIMED.
- **9. TAXES AND GOVERNMENTAL FEES.** Purchaser shall either pay or reimburse Provider for any and all taxes assessed on the generation, sale, delivery or consumption of electric energy produced by the System or the interconnection of the System to the Utility's electric distribution system, including property taxes on the System; <u>provided, however, Purchaser will not be required to pay or reimburse Provider for any taxes during periods when Provider fails to deliver electric energy to Purchaser due to the action or omission of Provider. For purposes of this <u>Section 4(d)</u>, "**Taxes**" means any federal, state and local ad valorem, property, occupation, generation, privilege, sales, use, consumption, excise, transaction, and other taxes, regulatory fees, surcharges or other similar charges, but shall not include any income taxes or similar taxes imposed on Provider's revenues due to the sale of energy under this Agreement, which shall be Provider's responsibility.</u>

10. FORCE MAJEURE.

- 10.1 Definition. "Force Majeure Event" means includes, but is not limited to any act or event that prevents the affected Party from performing its obligations in accordance with the Agreement, if such act or event is beyond the reasonable control, and not the result of the fault or negligence, of the affected Party and such Party had been unable to overcome such act or event with the exercise of due diligence (provided that the claiming Party shall not be required to suffer prejudice or use commercially unreasonable measures to remedy a Force Majeure event). A Force Majeure Event shall not be based on the economic hardship or financial condition of either Party. Force Majeure shall not be based on (i) Purchaser's inability economically to use or resell the Services purchased hereunder; or (ii) the loss or failure of Provider's supply.
- 10.2 Excused Performance. Except as otherwise specifically provided in the Agreement, neither Party shall be considered in breach of the Agreement or liable for any delay or failure to comply with the Agreement (other than the failure to pay amounts due hereunder), if and to the extent that such delay or failure is attributable to the occurrence of a Force Majeure Event; provided that the Party claiming relief under this Section 10 shall: (i) promptly notify the other Party in writing of the existence of, and details regarding, the Force Majeure Event; (ii) exercise all reasonable efforts necessary to minimize delay caused by such Force Majeure Event; (iii) notify the other Party in writing of the cessation or termination of said Force Majeure Event; and (iv) resume performance of its obligations hereunder as soon as practicable thereafter.

10.3 Termination in Consequence of Force Majeure Event. If a Force Majeure Event shall have occurred that has affected a Party's performance of its obligations hereunder and that has continued for a continuous period of one hundred eighty (180) days, then the Party not affected by Force Majeure shall be entitled to terminate the Agreement upon ninety (90) days' prior written notice to the affected Party. If at the end of such ninety (90) day period such Force Majeure Event shall still continue, the Agreement shall automatically terminate. Upon such termination for a Force Majeure Event, neither Party shall have any liability to the other (other than any such liabilities that have accrued prior to such termination), and the provisions of Section 2.2 (Early Termination) shall be inapplicable.

11. DEFAULT.

- 11.1 Provider Defaults and Purchaser Remedies. (a) Provider Defaults. The following events shall be defaults with respect to Provider (each, a "Provider Default"): (i) A Bankruptcy Event shall have occurred with respect to Provider; (ii) Provider fails to pay Purchaser any undisputed amount owed under the Agreement within ten (10) Business Days from receipt of notice from Purchaser of such past due amount; (iii) Provider breaches or fails to perform any material term of the Agreement (other than events that are otherwise specifically covered in this Section 11.1 as a separate Provider Default) and such breach or failure is not cured within thirty (30) days after Purchaser's written notice of such breach or failure; (iv) any representation or warranty made in this Agreement by Provider is not true and complete in any material respect when given or at any time during the Term; (v) INTENTIONALLY OMITTED; and (vi) Provider consolidates or amalgamates with, or merges with or into, or transfers all or substantially all of its assets to, another entity and, at the time of such consolidation, amalgamation, merger or transfer, the resulting, surviving or transferee entity fails to assume all the obligations of Provider under this Agreement to which it or its predecessor was a party by operation of law or pursuant to an agreement reasonably satisfactory to Purchaser.
- (b) Purchaser's Remedies. If a Provider Default described in Section 11.1(a) has occurred and is continuing, in addition to other remedies expressly provided herein, and subject to Section 12, Purchaser may terminate the Agreement and shall be entitled to liquidated damages in an amount equal to the positive difference, if any, of the Fair Market Value of the Services anticipated to be provided over the remaining Term, as determined by Purchaser in a commercially reasonable manner, and the Services Payments associated with such Services ("Termination Amount"). Upon termination of this Agreement in accordance with this Section, Purchaser shall promptly calculate the Termination Amount, if any, owed by Provider and submit an invoice to Provider for such amount. Provider shall pay such Termination Amount within ten (10) Business Days after receipt of such invoice in accordance with the payment directions specified by Purchaser In the event Provider, in good faith, disputes ("Termination Amount Payment Date"). Purchaser's calculation of the Termination Amount, the Parties shall then act in good faith to resolve any such dispute, including but not limited to utilizing a procedure substantially similar to that provided for in Section 2.4 in the event that the Parties are unable to resolve such dispute; provided however that if through such procedures the Termination Amount is determined to be equal to or greater than that calculated by Purchaser, Provider shall pay all costs associated with such procedure, or otherwise the Parties shall share such costs equally. (c) No Early Termination Fee. Section 2.2 of the Agreement shall not apply to any termination of the Agreement by Purchaser pursuant to this Section 11.1.

- 11.2 Purchaser Defaults and Provider's Remedies. (a) Purchaser Default. The following events shall be defaults with respect to Purchaser (each, a "Purchaser Default"): (i) A Bankruptcy Event shall have occurred with respect to Purchaser; (ii) Purchaser breaches or fails to perform any material term of the Agreement (other than events that are otherwise specifically covered in this Section 11.2 as a separate Purchaser Default) and such breach or failure is not cured within thirty (30) days after Provider's written notice of such breach or failure; (iii) Purchaser fails to pay Provider any undisputed amount due Provider under the Agreement within thirty (30) days from receipt of notice from Provider of such past due amount; (iv) any representation or warranty made in this Agreement by Purchaser is not true and complete in any material respect when given or at any time during the Term; and (v) Purchaser consolidates or amalgamates with, or merges with or into, or transfers all or substantially all of its assets to, another entity and, at the time of such consolidation, amalgamation, merger or transfer, the resulting, surviving or transferee entity fails to assume all the obligations of Purchaser under this Agreement to which it or its predecessor was a party by operation of law or pursuant to an agreement reasonably satisfactory to Provider.
- (b) Provider's Remedies. If a Purchaser Default described in Sections 11.2(a) has occurred and is continuing, in additional to other remedies expressly provided herein, and subject to Section 12, Provider may terminate this Agreement and upon such termination, and Provider shall be entitled to receive from Purchaser, as liquidated damages, an amount equal to the sum of (i) the Early Termination Fee pursuant to Section 2.2, (ii) any and all other amounts previously accrued under this Agreement and then owed by Purchaser to Provider, and (iii) removal costs as provided in Section 11.3.
- 11.3 Removal of System. Upon any termination of the Agreement pursuant to this Section 11, Provider will remove the System pursuant to Section 2.5 hereof at the expense of the defaulting Party.

12. LIMITATIONS OF LIABILITY.

12.1 THE PARTIES CONFIRM THAT THE EXPRESS REMEDIES AND MEASURES OF DAMAGES PROVIDED IN THIS AGREEMENT SATISFY THE ESSENTIAL PURPOSES HEREOF. FOR BREACH OF ANY PROVISION FOR WHICH AN EXPRESS REMEDY OR MEASURE OF DAMAGES IS PROVIDED, SUCH EXPRESS REMEDY OR MEASURE OF DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY, THE OBLIGOR'S LIABILITY SHALL BE LIMITED AS SET FORTH IN SUCH PROVISION AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAIVED. IF NO REMEDY OR MEASURE OF DAMAGES IS EXPRESSLY PROVIDED HEREIN, THE OBLIGOR'S LIABILITY SHALL BE LIMITED TO DIRECT ACTUAL DAMAGES ONLY, SUCH DIRECT ACTUAL DAMAGES SHALL BE THE SOLE AND EXCLUSIVE DAMAGES. UNLESS EXPRESSLY HEREIN PROVIDED, NEITHER PARTY SHALL BE LIABLE FOR CONSEQUENTIAL, INCIDENTAL, PUNITIVE, EXEMPLARY INDIRECT DAMAGES, LOST PROFITS OR OTHER BUSINESS INTERRUPTION DAMAGES, BY STATUTE, IN TORT OR CONTRACT OR OTHERWISE. . TO THE EXTENT ANY DAMAGES REQUIRED TO BE PAID HEREUNDER ARE LIQUIDATED, THE PARTIES ACKNOWLEDGE THAT THE DAMAGES ARE DIFFICULT OR IMPOSSIBLE TO DETERMINE, OR OTHERWISE OBTAINING AN ADEQUATE REMEDY IS INCONVENIENT AND THE DAMAGES CALCULATED HEREUNDER CONSTITUTE A REASONABLE APPROXIMATION OF THE HARM OR LOSS.

NOTWITHSTANDING THE FOREGOING, PROVIDER SHALL REMAIN LIABLE, WITHOUT MONETARY LIMITATION, FOR DIRECT DAMAGES FOR PERSONAL INJURY, DEATH OR DAMAGE TO REAL PROPERTY OR TANGIBLE PERSONAL PROPERTY ATTRIBUTABLE TO THE NEGLIGENCE OR OTHER TORT OF THE PROVIDER, ITS OFFICERS, EMPLOYEES OR AGENTS.

13. ASSIGNMENT.

13.1 Assignment by Purchaser. Purchaser may assign or transfer this Agreement to any other New York State agency or public benefit corporation that is the successor in interest of the University and that is backed by the full faith and credit of the State of New York without the Provider's prior consent, provided that (a) Purchaser shall promptly notify Seller of any such assignment in writing, (b) such assignee shall assume all of Purchaser's obligations under the Agreement in writing; and (c) such assignment shall in no event compromise the tax treatment of this Agreement or the Provider. Any assignment that does not comply with the forgoing shall be invalid.

14. NOTICES.

- 14.1 Notice Addresses. Unless otherwise provided in the Agreement, all notices and communications concerning the Agreement shall be in writing and addressed to the other Party (or Financing Party, as the case may be) at the addresses set forth in Schedule 6, or at such other address as may be designated in writing to the other Party from time to time.
- 14.2 Notice. Unless otherwise provided herein, any notice provided for in the Agreement shall be hand delivered, sent by registered or certified U.S. Mail, postage prepaid, or by commercial overnight delivery service, or transmitted by facsimile and shall be deemed delivered to the addressee or its office when received at the address for notice specified above if received during normal business hours on a Business Day or otherwise such notice shall be deemed delivered on the next Business Day.
- 14.3 Address for Invoices. All invoices under the Agreement shall be sent to the address provided by Purchaser. Invoices shall be sent by regular first class mail postage prepaid.

15. CONFIDENTIALITY.

15.1 Confidentiality Obligation. If either Party provides confidential information, including business plans, strategies, financial information, proprietary, patented, licensed, copyrighted or trademarked information, and/or technical information regarding the financing, design, operation and maintenance of the System or of Purchaser's business ("Confidential Information") to the other or, if in the course of its performance of this Agreement, or if in the course of negotiating this Agreement a Party learns Confidential Information regarding the facilities or plans of the other, the receiving Party shall (a) protect the Confidential Information from disclosure to third parties with the same degree of care accorded its own confidential and proprietary information, and (b) refrain from using such Confidential Information, except in the negotiation and performance of the Agreement. Notwithstanding the above, a Party may with the consent of the other Party (such consent not to be unreasonably conditioned, delayed or withheld, provide such

Confidential Information to its officers, directors, members, managers, employees, agents, attorneys, accountants and consultants, Affiliates, lenders, and potential assignees of the Agreement or acquirers of Provider or its Affiliates (provided and on condition that such potential assignees be bound by a written agreement restricting use and disclosure of Confidential Information) (collectively, "Representatives"), in each case whose access is reasonably necessary. Each such recipient of Confidential Information shall be informed by the Party disclosing Confidential Information of its confidential nature and shall be directed to treat such information confidentially and shall agree to abide by these provisions. In any event, each Party shall be liable (with respect to the other Party) for any breach of this provision by its Representatives. All Confidential Information shall remain the property of the disclosing Party and shall be returned to the disclosing Party or destroyed after the receiving Party's need for it has expired or upon the request of the disclosing Party.

- 15.2 Permitted Disclosures. Notwithstanding any other provision herein, neither Party shall be required to hold confidential any information that: (a) becomes publicly available other than through the receiving Party or its Representatives; (b) is required to be disclosed by a Governmental Authority, under Applicable Law or pursuant to a validly issued subpoena or required filing, but a receiving Party subject to any such requirement shall promptly notify the disclosing Party of such requirement; (c) is independently developed by the receiving Party; or (d) becomes available to the receiving Party without restriction from a third party under no obligation of confidentiality. The Parties acknowledge that this Agreement is subject to the New York State Freedom of Information Law ("FOIL") as set forth in Article 6 of the New York State Public Officers Law, and that only Provider's proprietary information that satisfies the requirements of § 87(2)(d) of the Public Officers Law shall be excepted from disclosure thereunder.
- 15.3 Goodwill and Publicity. Neither Party shall use the name, trade name, service mark, or trademark of the other Party in any promotional or advertising material without the prior written consent of such other Party. The Parties shall coordinate and cooperate with each other when making public announcements related to the execution and existence of the Agreement, and each Party shall have the right to promptly review, comment upon, and approve any publicity materials, press releases, or other public statements by the other Party that refer to, or that describe any aspect of, the Agreement; provided that no such publicity releases or other public statements (except for filings or other statements or releases as may be required by Applicable Law) shall be made by either Party without the prior written consent of the other Party. At no time will either Party acquire any rights whatsoever to any trademark, trade name, service mark, logo or other intellectual property right belonging to the other Party. Notwithstanding the foregoing, Purchaser agrees that Provider may, at its sole discretion, take photographs of the installation process of the System and/or the completed System, and Provider shall be permitted to use such images (regardless of media) in its marketing efforts, including but not limited to use in brochures, advertisements, websites and news outlet or press release articles. The images shall not include any identifying information without Purchaser's prior written authorization.
- 15.4 Enforcement of Confidentiality Obligation. Each Party agrees that the disclosing Party would be irreparably injured by a breach of this Article by the receiving Party or its Representatives or other Person to whom the receiving Party discloses Confidential Information of the disclosing Party and that the disclosing Party is entitled to seek equitable relief, including

injunctive relief and specific performance, in the event of any breach of the provisions of this Article. To the fullest extent permitted by Applicable Law, such remedies shall not be deemed to be the exclusive remedies for a breach of this Article, but shall be in addition to all other remedies available at law or in equity.

16. INDEMNITY.

16.1 Provider's Indemnity. Subject to Section 12, Provider agrees that it shall indemnify and hold harmless Purchaser, its permitted successors and assigns and their respective directors, officers, members, shareholders and employees (collectively, the "Purchaser Indemnified Parties") from and against any and all Losses incurred by the Purchaser Indemnified Parties to the extent arising from or out of the following: (a) any claim by a third party for or arising out of any injury to or death of any Person or loss or damage to property of any Person to the extent arising out of Provider's negligence or willful misconduct or (b) any claim by a third party regarding any infringement of patents or the improper use of other proprietary rights by Provider or its employees or representatives that may occur in connection with the performance of the Installation Work, System Operations, and the ownership and use of the System. Provider shall not, however, be required to reimburse or indemnify any Purchaser Indemnified Party for any Loss to the extent such Loss is due to the negligence or willful misconduct of any Purchaser Indemnified Party.

16.2 Purchaser's Indemnity. Subject to the availability of lawful appropriations and consistent with Section 8 of the New York State Court of Claims Act, Purchaser shall hold Provider harmless from and indemnify it for any final judgment of a court of competent jurisdiction to the extent attributable to the negligence of Purchaser or of its officers or employees when acting within the course and scope of their employment. In accordance with § 9-103 (2) of New York State Energy Law, this Agreement shall be deemed executory only to the extent monies are appropriated and available for the purpose of the contract, and no liability on account herefor shall be incurred beyond the amount of such monies. It is understood that neither this contract nor any representation by any public employee or officer creates any legal or moral obligation to request, appropriate or make available monies for the purpose of the contract.

17. INSURANCE.

17.1 Generally. Provider shall each maintain the following insurance coverages in full force and effect throughout the Term either through insurance policies or acceptable self-insured retentions: (a) Workers' Compensation Insurance as may be from time to time required under applicable federal and state law, (b) Commercial General Liability Insurance with limits of not less than \$2,000,000 general aggregate, \$2,000,000 per occurrence, and (c) automobile insurance with a combined single limit of not less than \$1,000,000. Additionally, Provider shall carry adequate property loss insurance on the System. The amount and terms of insurance coverage will be determined at Provider's sole discretion.

17.2 Certificates of Insurance. Upon Purchaser's request, Provider shall furnish current certificates evidencing that the insurance required under Section 17.1 is being maintained. The

insurance policy provided hereunder shall contain a provision whereby the insured agrees to give the Purchaser thirty (30) days' written notice before the insurance is cancelled or materially altered, ten (10) days' notice in case of a termination or cancellation due to non-payment of premium. Purchaser is self-insured for its general liability exposure, workers compensation and automobile liability. In accordance with Article II of the New York Court of Claims Act, it shall be liable for damage or injury arising from negligent actions caused by an officer of the State University or its employees acting within the scope of their respective employment, Purchaser's self-insured funds could be drawn down if: (i) a judgment of liability is made under the provisions of the New York Court Claims Act, or in the case of small claims, an acknowledgement of liability has been made by the Attorney General and the Comptroller of the State of New York. The State's obligation with respect to any such claims are subject to the availability of lawful appropriations, as required by Section 41 of the New York State Finance Law, Purchaser agrees to notify Provider of any material change of the status of Purchaser's self-insurance with at least thirty (30) consecutive days; prior written notice.

- 17.3 Additional Insureds. Provider' Commercial General Liability insurance policy shall be written on an occurrence basis and shall include the Purchaser as an additional insured as its interest may appear.
- 17.4 Insurer Qualifications. All insurance maintained hereunder shall be maintained with companies authorized by the New York Department of Financial Services to issue insurance in the State of New York ("admitted carriers) with an A.M. Best Company rating of "A-" or better, or having a parent company's debt to policyholder surplus ratio of 1:1.

18. MISCELLANEOUS.

18.1 INTENTIONALLY OMITTED.

- 18.2 Amendments. This Agreement may only be amended, modified or supplemented by an instrument in writing executed by duly authorized representatives of Provider and Purchaser and approved by the New York State Attorney General and Office of Comptroller.
- 18.3 Industry Standards. Unless expressly defined herein, words having well-known technical or trade meanings shall be so construed.

18.4 Intentionally Omitted

- 18.5 Limited Effect of Waiver. The failure of Provider or Purchaser to enforce any of the provisions of the Agreement, or the waiver thereof, shall not be construed as a general waiver or relinquishment on its part of any such provision, in any other instance or of any other provision in any instance.
- 18.6 Survival. The obligations under Sections 2.2 (Early Termination), 2.5 (Removal of System), Section 7 (General Covenant), Sections 7.2(d), (e), (f) and (g) (Purchaser Covenants), Section 8.3 (Exclusion of Warranties), Article 9 (Taxes and Governmental Fees), Article 12 (Limitation of Liability), Article 14 (Notices), Article 15 (Confidentiality), Article 16 (Indemnity), Article 17 (Insurance), Article 18 (Miscellaneous), or pursuant to other provisions of this Agreement that,

by their sense and context, are intended to survive termination of this Agreement shall survive the expiration or termination of this Agreement for any reason.

- 18.7 Governing Law. All disputes arising out of the performance or non-performance under this Agreement shall be construed in accordance with the laws of the State of New York, without regard to principles of conflicts of law other than New York General Obligations Law § 5-1401. The Parties agree that any dispute hereunder will be litigated in a court of competent jurisdiction in New York.
- 18.8 Relationship to the Parties. Nothing contained in this Agreement shall be construed to create an association, joint venture, partnership or any other type of entity or relationship between Purchase and Provider, or between either or both of them and any other Party.
- 18.9 Third-Party Beneficiaries. Except as expressly provided herein to the contrary, this Agreement is intended solely for the benefit of the Parties thereto, and nothing therein will be construed to create any duty to, or standard of care with reference to, or any liability to, any person not a Party thereto.
- 18.10 Counterparts. This Agreement may be executed in any number of counterparts, each of which shall be an original, but all of which together shall constitute one instrument.
- 18.11 Forward Contract; Service Contract. The Parties acknowledge and agree that this Agreement constitutes a "forward contract" within the meaning of the United States Bankruptcy Code. The payment of the Termination Amount or Early Termination Fee made or to be made by one Party to the other Party under this Agreement constitutes a "settlement payment" and/or a "transfer" under the United States Bankruptcy Code. This agreement constitutes a "master netting agreement" and each party is a "master netting agreement participant" within the meaning of the United States Bankruptcy Code. The Parties intend this Agreement to be a "service contract" within the meaning of Section 7701(e)(3) of the Internal Revenue Code of 1986. Purchaser will not take the position on any tax return or in any other filings suggesting that it is anything other than a purchase of electricity from the System.
- 18.12 Iran Divestment. By entering into this Agreement, Provider certifies that it is not on the "Entities Determined To Be Non-Responsive Bidders/Offerers Pursuant to The New York State Iran Divestment Act of 2012" list ("Prohibited Entities List") posted on the New York Office of General Services website at:

http://www.ogs.ny.gov/about/regs/docs/ListofEntities.pdf and further certifies that in the performance of this Agreement, it will not utilize any subcontractor that is identified on the Prohibited Entities List. Provider Provider agrees that should it seek to renew or extend this Agreement, it must provide the same certification at the time of renewal or extension. Provider also agrees that any proposed Assignee of the Agreement will be required to certify that it is not on the Prohibited Entities List before SUNY may approve any request for assignment of this Agreement. During the term of the Agreement, should SUNY receive information that a person (as defined in State Finance Law §165-a) is in violation of the above-referenced certifications, SUNY will review such information and offer the person an opportunity to respond. If the person fails to demonstrate that it has ceased its engagement in the investment activity which is in violation of the Act within 90 days after the determination of such violation, then SUNY shall

take such action as may be appropriate and provided for by law, rule, or contract, including, but not limited to, seeking compliance, recovering damages, or declaring the Provider in default. SUNY reserves the right to reject any request for renewal, extension, or assignment for an entity that appears on the Prohibited Entities List prior to the renewal, extension, or assignment of this Agreement, and to pursue a responsibility review with Provider should it appear on the Prohibited Entities List hereafter.

- 18.13 Grounds to Termination for Provider's Responsibility.
- (a) <u>General Responsibility.</u> Provider shall at all times during the Term remain Responsible Provider agrees, if requested by SUNY Chancellor or his or her designee, to present evidence of its continuing legal authority to do business in New York State, integrity, experience, ability, prior performance, and organizational and financial capacity
- (b) <u>Suspension of Work (for Non-Responsibility)</u>. The SUNY Chancellor or his or her designee, in his or her sole discretion, reserves the right to suspend any or all activities under this Contract, at any time, In the event of such suspension, Provider will be given written notice outlining the particulars of such suspension. Upon issuance of such notice, Provider must Contract activity may resume at such time as the SUNY Chancellor or his or her designee issues a written notice authorizing a resumption of performance under the Contract.
- (c) Termination (for Non-Responsibility). Upon written notice to Provider, and a reasonable opportunity to be heard with appropriate SUNY officials or staff, the Contract may be terminated by SUNY Chancellor or his or her designee at Provider's expense where Provider is determined by the SUNY Chancellor or his or her designee to be non-responsible. In such event, the SUNY Chancellor or his or her designee may complete the contractual requirements in any manner he or she may deem advisable and pursue available legal or equitable remedies for breach.

IN WITNESS WHEREOF and in confirmation of their consent to the terms and conditions contained in this Agreement and intending to be legally bound hereby, Provider and Purchaser have executed this Agreement as of the Effective Date.

PROVIDER:
By:
Name:
Title:
Date:
PURCHASER: SUNY Purchase
By:
Name:
Title:
Date:

Standard Contract Clauses

EXHIBIT A

State University of New York

December 13, 2012

The parties to the attached contract, license, lease, amendment or other agreement of any kind (hereinafter, "contract") agree to be bound by the following clauses which are hereby made a part of the contract (the word "Contractor" herein refers to any party other than the State, whether a Contractor, licensee, lessor, lessee or any other party):

- 1. EXECUTORY CLAUSE. In accordance with Section 41 of the State Finance Law, the State shall have no liability under this contract to the Contractor or to anyone else beyond funds appropriated and available for this contract.
- 2. PROHIBITION AGAINST ASSIGNMENT Except for the assignment of its right to receive payments subject to Article 5-A of the State Finance Law, the Contractor selected to perform the services herein are prohibited in accordance with Section 138 of the State Finance Law from assigning, transferring, conveying, subletting or otherwise disposing of its rights, title or interest in the contract without the prior written consent of SUNY and attempts to do so are null and void. Notwithstanding the foregoing, SUNY may, with the concurrence of the New York Office of State Comptroller, waive prior written consent of the assignment, transfer, conveyance, sublease or other disposition of a contract let pursuant to Article XI of the State Finance Law if the assignment, transfer, conveyance, sublease or other disposition is due to a reorganization, merger or consolidation of Contractor's its business entity or enterprise and Contractor so certifies to SUNY. SUNY retains the right, as provided in Section 138 of the State Finance Law, to accept or reject an assignment, transfer, conveyance, sublease or other disposition of the contract, and to require that any Contractor demonstrate its
- 3. COMPTROLLER'S APPROVAL. (a) In accordance with Section 112 of the State Finance Law, Section 355 of New York State Education Law, and 8 NYCRR 316, Comptroller's approval is not required for the following contracts: (i) materials; (ii) equipment and supplies, including computer equipment; (iii) motor vehicles; (iv) construction; (v) construction-related services; (vi) printing; and (vii) goods for State University health care facilities, including contracts for goods made with joint or group purchasing arrangements.

responsibility to do business with SUNY.

- (b) Comptroller's approval is required for the following contracts: (i) contracts for services not listed in Paragraph (3)(a) above made by a State University campus or health care facility certified by the Vice Chancellor and Chief Financial Officer, if the contract value exceeds \$250,000; (ii) contracts for services not listed in Paragraph (3)(a) above made by a State University campus not certified by the Vice Chancellor and Chief Financial Officer, if the contract value exceeds \$50,000; (iii) contracts for services not listed in Paragraph (3)(a) above made by health care facilities not certified by the Vice Chancellor and Chief Financial Officer, if the contract value exceeds \$75,000; (iv) contracts whereby the State University agrees to give something other than money, when the value or reasonably estimated value of such consideration exceeds \$10,000; (v) contracts for real property transactions if the contract value exceeds \$50,000; (vi) all other contracts not listed in Paragraph 3(a) above, if the contract value exceeds \$50,000, e.g. SUNY acquisition of a business and New York State Finance Article 11-B contracts and (vii) amendments for any amount to contracts not listed in Paragraph (3)(a) above, when as so amended, the contract exceeds the threshold amounts stated in Paragraph (b) herein. However, such pre-approval shall not be required for any contract established as a centralized contract through the Office of General Services or for a purchase order or other transaction issued under such centralized contract.
- (c) Any contract that requires Comptroller approval shall not be valid, effective or binding upon the State University until it has been approved by the Comptroller and filed in the Comptroller's office.

- 4. WORKERS' COMPENSATION BENEFITS. In accordance with Section 142 of the State Finance Law, this contract shall be void and of no force and effect unless the Contractor shall provide and maintain coverage during the life of this contract for the benefit of such employees as are required to be covered by the provisions of the Workers' Compensation Law.
- 5. NON-DISCRIMINATION REQUIREMENTS.
- To the extent required by Article 15 of the Executive Law (also known as the Human Rights Law) and all other State and Federal statutory and constitutional non-discrimination provisions, the Contractor will not discriminate against any employee or applicant for employment because of race, creed, color, sex, national origin, sexual orientation, age, disability, genetic predisposition or carrier status, or marital status. Furthermore, in accordance with Section 220-e of the Labor Law, if this is a contract for the construction, alteration or repair of any public building or public work or for the manufacture, sale or distribution of materials, equipment or supplies, and to the extent that this contract shall be performed within the State of New York, Contractor agrees that neither it nor its subcontractors shall, by reason of race, creed, color, disability, sex, or national origin: (a) discriminate in hiring against any New York State citizen who is qualified and available to perform the work; or (b) discriminate against or intimidate any employee hired for the performance of work under this contract. If this is a building service contract as defined in Section 230 of the Labor Law, then, in accordance with Section 239 thereof, Contractor agrees that neither it nor its subcontractors shall by reason of race, creed, color, national origin, age, sex or disability: (a) discriminate in hiring against any New York State citizen who is qualified and available to perform the work; or (b) discriminate against or intimidate any employee hired for the performance of work under this contract. Contractor is subject to fines of \$50.00 per person per day for any violation of Section 220-e or Section 239 as well as possible termination of this contract and forfeiture of all moneys due hereunder for a second or subsequent violation
- 6. WAGE AND HOURS PROVISIONS. If this is a public work contract covered by Article 8 of the Labor Law or a building service contract covered by Article 9 thereof, neither Contractor's employees nor the employees of its subcontractors may be required or permitted to work more than the number of hours or days stated in said statutes, except as otherwise provided in the Labor Law and as set forth in prevailing wage and supplement schedules issued by the State Labor Department. Furthermore, Contractor and its subcontractors must pay at least the prevailing wage rate and pay or provide the prevailing supplements, including the premium rates for overtime pay, as determined by the State Labor Department in accordance with the Labor Law. Additionally, effective April 28, 2008, if this is a public work contract covered by Article 8 of the Labor Law, the Contractor understands and agrees that the filing of payrolls in a manner consistent with Subdivision 3-a of Section 220 of the Labor Law shall be a condition precedent to payment by SUNY of any SUNYapproved sums due and owing for work done upon the project.
- 7. NON-COLLUSIVE BIDDING CERTIFICATION. In accordance with Section 139-d of the State Finance Law, if this contract was awarded based on the submission of competitive bids, Contractor affirms, under penalty of perjury, and each person signing on behalf of Contractor, and in the case of a joint bid each party thereto certifies as to its own organization, under penalty of perjury, that to the best of its knowledge and belief that its bid was arrived at independently and without collusion aimed at restricting competition. Contractor further affirms that,

at the time Contractor submitted its bid, an authorized and responsible person executed and delivered it to SUNY a non-collusive bidding certification on Contractor's behalf.

8. INTERNATIONAL BOYCOTT PROHIBITION.

- In accordance with Section 220-f of the Labor Law and Section 139-h of the State Finance Law, if this contract exceeds \$5,000, the Contractor agrees, as a material condition of the contract, that neither the Contractor nor any substantially owned or affiliated person, firm, partnership or corporation has participated, is participating, or shall participate in an international boycott in violation of the federal Export Administration Act of 1979 (50 USC App. Sections 2401 et seq.) or regulations thereunder. If such Contractor, or any of the aforesaid affiliates of Contractor, is convicted or is otherwise found to have violated said laws or regulations upon the final determination of the United States Commerce Department or any other appropriate agency of the United States subsequent to the contract's execution, such contract, amendment or modification thereto shall be rendered forfeit and void. The Contractor shall so notify the State Comptroller within five (5) business days of such conviction, determination or disposition of appeal (2 NYCRR 105.4).
- 9. **SET-OFF RIGHTS.** The State shall have all of its common law, equitable and statutory rights of set-off. These rights shall include, but not be limited to, the State 's option to withhold for the purposes of set-off any moneys due to the Contractor under this contract up to any amounts due and owing to the State with regard to this contract, any other contract with any State department or agency, including any contract for a term commencing prior to the term of this contract, plus any amounts due and owing to the State for any other reason including, without limitation, tax delinquencies or monetary penalties relative thereto. The State shall exercise its set-off rights in accordance with normal State practices including, in cases of setoff pursuant to an audit, the finalization of such audit by the State, its representatives, or the State Comptroller.
- 10. RECORDS. The Contractor shall establish and maintain complete and accurate books, records, documents, accounts and other evidence directly pertinent to performance under this (hereinafter, collectively, "the Records"). The Records must be kept for the balance of the calendar year in which they were made and for six (6) additional years thereafter. The State Comptroller, the Attorney General and any other person or entity authorized to conduct an examination, as SUNY and its representatives and entities involved in this contract, shall have access to the Records during normal business hours at an office of the Contractor within the State of New York or, if no such office is available, at a mutually agreeable and reasonable venue within the State, for the term specified above for the purposes of inspection, auditing and copying. SUNY shall take reasonable steps to protect from public disclosure any of the Records which are exempt from disclosure under Section 87 of the Public Officers Law (the "Statute") provided that: (i) the Contractor shall timely inform an appropriate SUNY official, in writing, that said Records should not be disclosed; and (ii) said Records shall be sufficiently identified; and (iii) designation of said Records as exempt under the Statute is reasonable. Nothing contained herein shall diminish, or in any way adversely affect, SUNY's or the State's right to discovery in any pending or future litigation.

11. **IDENTIFYING INFORMATION AND PRIVACY NOTIFICATION.**

Identification Number(s). Every invoice or New York State Claim for Payment submitted to the State

University of New York by a payee, for payment for the sale of goods or services or for transactions (e.g., leases, easements, licenses, etc.) related to real or personal property must include the payee's identification number. The number is any or all of the following: (i) the payee's Federal employer identification number, (ii) the payee's Federal social security number, and/or (iii) the payee's Vendor Identification Number assigned by the Statewide Financial System. Failure to include such number or numbers may delay payment. Where the payee does not have such number or numbers, the payee, on its invoice or Claim for Payment, must give the reason or reasons why the payee does not have such number or numbers.

(b) Privacy Notification. (1) The authority to request the above personal information from a seller of goods or services or a lessor of real or personal property, and the authority to maintain such information, is found in Section 5 of the State Tax Law. Disclosure of this information by the seller or lessor to the State University of New York is mandatory. The principal purpose for which the information is collected is to enable the State to identify individuals, businesses and others who have been delinquent in filing tax returns or may have understated their tax liabilities and to generally identify persons affected by the taxes administered by the Commissioner of Taxation and Finance. The information will be used for tax administration purposes and for any other purpose authorized by law. (2) The personal information is requested by the purchasing unit of the State University of New York contracting to purchase the goods or services or lease the real or personal property covered by this contract or lease. The information is maintained in the Statewide Financial System by the Vendor Management Unit within the Bureau of State Expenditures, Office of the State Comptroller, 110 State Street, Albany, New York

12. EQUAL EMPLOYMENT OPPORTUNITIES FOR MINORITIES AND WOMEN.

(a) In accordance with Section 312 of the Executive Law and 5 NYCRR 143, if this contract is: (i) a written agreement or purchase order instrument, providing for a total expenditure in excess of \$25,000.00, whereby a contracting agency is committed to expend or does expend funds in return for labor, services, supplies, equipment, materials or any combination of the foregoing, to be performed for, or rendered or furnished to the contracting agency; or (ii) a written agreement in excess of \$100,000.00 whereby a contracting agency is committed to expend or does expend funds for the acquisition, construction, demolition, replacement, major repair or renovation of real property and improvements thereon; or (iii) a written agreement in excess of \$100,000.00 whereby the owner of a State assisted housing project is committed to expend or does expend funds for the acquisition, construction, demolition, replacement, major repair or renovation of real property and improvements thereon for such project, then the following shall apply and by signing this agreement the Contractor certifies and affirms that it is Contractor's equal employment opportunity policy that:

- (1) The Contractor will not discriminate against employees or applicants for employment because of race, creed, color, national origin, sex, age, disability or marital status, and will undertake or continue existing programs of affirmative action to ensure that minority group members and women are afforded equal employment opportunities without discrimination. Affirmative action shall mean recruitment, employment, job assignment, promotion, upgradings, demotion, transfer, layoff, or termination and rates of pay or other forms of compensation;
- (2) at SUNY's request, Contractor shall request each employment agency, labor union, or authorized representative of workers with which it has a collective bargaining or other agreement or understanding, to furnish a written statement that such employment agency, labor union or representative will not discriminate on the basis of race, creed, color, national origin, sex, age, disability or marital status and that such union or representative will affirmatively cooperate in the implementation of the Contractor's obligations herein; and
- (3) Contractor shall state, in all solicitations or advertisements for employees, that, in the performance of

the State contract, all qualified applicants will be afforded equal employment opportunities without discrimination because of race, creed, color, national origin, sex, age, disability or marital status.

- (b) Contractor will include the provisions of "1", "2" and "3", above, in every subcontract over \$25,000.00 for the construction, demolition, replacement, major repair, renovation, planning or design of real property and improvements thereon (the "Work") except where the Work is for the beneficial use of the Contractor. Section 312 does not apply to: (i) work, goods or services unrelated to this contract; or (ii) employment outside New York State. The State shall consider compliance by a Contractor or subcontractor with the requirements of any federal law concerning equal employment opportunity which effectuates the purpose of this section. SUNY shall determine whether the imposition of the requirements of the provisions hereof duplicate or conflict with any such federal law and if such duplication or conflict exists, SUNY shall waive the applicability of Section 312 to the extent of such duplication or conflict. Contractor will comply with all duly promulgated and lawful rules and regulations of the Department of Economic Development's Division of Minority and Women's Business Development pertaining hereto.
- 13. **CONFLICTING TERMS.** In the event of a conflict between the terms of the contract (including any and all attachments thereto and amendments thereof) and the terms of this Exhibit A, the terms of this Exhibit A shall control.
- 14. **GOVERNING LAW.** This contract shall be governed by the laws of the State of New York except where the Federal supremacy clause requires otherwise.
- 15. **LATE PAYMENT.** Timeliness of payment and any interest to be paid to Contractor for late payment shall be governed by Article 11-A of the State Finance Law to the extent required by law.
- 16. **NO ARBITRATION.** Disputes involving this contract, including the breach or alleged breach thereof, may not be submitted to binding arbitration (except where statutorily authorized) but must, instead, be heard in a court of competent jurisdiction of the State of New York.
- 17. SERVICE OF PROCESS. In addition to the methods of service allowed by the State Civil Practice Law & Rules ("CPLR"), Contractor hereby consents to service of process upon it by registered or certified mail, return receipt requested. Service hereunder shall be complete upon Contractor's actual receipt of process or upon the State's receipt of the return thereof by the United States Postal Service as refused or undeliverable. Contractor must promptly notify the State, in writing, of each and every change of address to which service of process can be made. Service by the State to the last known address shall be sufficient. Contractor will have thirty (30) calendar days after service hereunder is complete in which to respond.
- PROHIBITION ON PURCHASE OF TROPICAL HARDWOODS. The Contractor certifies and warrants that all wood products to be used under this contract award will be in accordance with, but not limited to, the specifications and provisions of State Finance Law §165 (Use of Tropical Hardwoods), which prohibits purchase and use of tropical hardwoods, unless specifically exempted, by the State or any governmental agency or political subdivision or public benefit corporation. Qualification for an exemption under this law will be the responsibility of the contractor to establish to meet with the approval of the State. In addition, when any portion of this contract involving the use of woods, whether supply or installation, is to be performed by any subcontactor, the prime Contractor will indicate and certify in the submitted bid proposal that the subcontractor has been informed and is in compliance with specifications and provisions regarding use of tropical hardwoods as detailed in Section 165 of the State Finance Law. Any such use must meet with the approval of the State, otherwise, the bid may not be considered responsive. Under bidder certification, proof of qualification for

Solar Purchase Power Agreement Template
vill be exemption will be the responsibility of the Contractor
to meet with the approval of the State.

19. MacBRIDE FAIR EMPLOYMENT PRINCIPLES. In accordance with the MacBride Fair Employment Principles (Chapter 807 of the Laws of 1992) , the Contractor hereby stipulates that Contractor and any individual or legal entity in which the Contractor holds a ten percent or greater ownership interest and any individual or legal entity that holds a ten percent or greater ownership interest in the Contractor either (a) have no business operations in Northern Ireland, or (b) shall take lawful steps in good faith to conduct any business operations in Northern Ireland in accordance with the MacBride Fair Employment Principles (as described in Section 165(5) of the State Finance Law), and shall permit independent monitoring of compliance with such principles.

20. OMNIBUS PROCUREMENT ACT OF 1992.

It is the policy of New York State to maximize opportunities for the participation of New York State business enterprises, including minority and womenowned business enterprises as bidders, subcontractors and suppliers on its procurement contracts.

Information on the availability of New York State subcontractors and suppliers is available from:

NYS Department of Economic Development Division for Small Business 30 South Pearl St., 7th Floor Albany, NY 12245 Tel: 518-292-5100 Fax: 518-292-5884

A directory of certified minority and womenowned business enterprises is available from:

email: opa@esd.ny.gov

NYS Department of Economic Development Division of Minority and Women's Business Development 633 Third Avenue New York, NY 10017 212-803-2414

email: mwbecertification@esd.ny.gov http://esd.ny.gov/MWBE/directorySearch.html

The Omnibus Procurement Act of 1992 requires that by signing this bid proposal or contract, as applicable, Contractors certify that whenever the total bid amount is greater than \$1 million:

- (a) The Contractor has made reasonable efforts to encourage the participation of New York State Business Enterprises as suppliers and subcontractors, including certified minority and women-owned business enterprises, on this project, and has retained the documentation of these efforts to be provided upon request to SUNY:
- (b) The Contractor has complied with the Federal Equal Employment Opportunity Act of 1972 (P.L. 92-261), as amended;
- (c) The Contractor agrees to make reasonable efforts to provide notification to New York State residents of employment opportunities on this project through listing any such positions with the Job Search Division of the New York State Department of Labor, or providing such notification in such manner as is consistent with existing collective bargaining contracts or agreements. The contractor agrees to document these efforts and to provide said documentation to the State upon request; and
- (d) The Contractor acknowledges notice that SUNY may seek to obtain offset credits from foreign countries as a result of this contract and agrees to cooperate with SUNY in these efforts.

21. RECIPROCITY AND SANCTIONS

PROVISIONS. Bidders are hereby notified that if their principal place of business is located in a country, nation, province, state or political subdivision that penalizes New York State vendors, and if the goods or services they offer will be substantially produced or performed outside New York State, the Omnibus Procurement Act of 1994 and 2000 amendments (Chapter 684 and Chapter 383, respectively) require

that they be denied contracts which they would otherwise obtain. Contact the NYS Department of Economic Development, Division for Small Business, 30 South Pearl Street, Albany, New York 12245, for a current list of jurisdictions subject to this provision.

- 22. COMPLIANCE WITH NEW YORK STATE INFORMATION SECURITY BREACH AND NOTIFICATION ACT. Contractor shall comply with the provisions of the New York State Information Security Breach and Notification Act (General Business Law Section 899-aa; State Technology Law Section 208).
- 23. COMPLIANCE WITH CONSULTANT DISCLOSURE LAW If this is a contract for consulting services, defined for purposes of this requirement to include analysis, evaluation, research, training, data processing, computer

programming, engineering, environmental health and mental health services, accounting, auditing, paralegal, legal or similar services, then in accordance with Section 163(4-g) of the State Finance Law, the Contractor shall timely, accurately and properly comply with the requirement to submit an annual

employment report for the contract to SUNY, the Department of Civil Service and the State Comptroller.

- 24. PURCHASES OF APPAREL AND SPORTS EQUIPMENT. In accordance with State Finance Law Section 165(7), SUNY may determine that a bidder on a contract for the purchase of apparel or sports equipment is not a responsible bidder as defined in State Finance Law Section 163 based on (a) the labor standards applicable to the manufacture of the apparel or sports equipment, including employee compensation, working conditions, employee rights to form unions and the use of child labor; or (b) bidder's failure to provide information sufficient for SUNY to determine the labor conditions applicable to the manufacture of the apparel or sports equipment.
- 25. **PROCUREMENT LOBBYING**. To the extent this agreement is a "procurement contract" as defined by State Finance Law Sections 139-j and 139-k, by signing this agreement the contractor certifies and affirms that all disclosures made in accordance with State Finance Law Sections 139-j and 139-k are complete, true and accurate. In the event such certification is found to be intentionally false or

intentionally incomplete, the State may terminate the agreement by providing written notification to the Contractor in accordance with the terms of the agreement.

Solar Purchase Power Agreement Template

26. CERTIFICATION OF REGISTRATION TO COLLECT SALES AND COMPENSATING USE TAX BY CERTAIN STATE CONTRACTORS, AFFILIATES AND SUBCONTRACTORS. To the extent this agreement is a contract as defined by Tax Law Section 5-a, if the Contractor fails to make the certification required by Tax Law Section 5-a or if during the term of the contract, the Department of Taxation and Finance or SUNY discovers that the certification, made under penalty of perjury, is false, then such failure to file or false certification shall be a material breach of this contract and this contract may be terminated, by providing written notification to the Contractor in accordance with the terms of the agreement, if SUNY determines that such action is in the best interests of the State.

THE FOLLOWING PROVISIONS SHALL APPLY ONLY TO THOSE CONTRACTS TO WHICH A HOSPITAL OR OTHER HEALTH SERVICE FACILITY IS A PARTY

- 27. Notwithstanding any other provision in this contract, the hospital or other health service facility remains responsible for insuring that any service provided pursuant to this contract complies with all pertinent provisions of Federal, state and local statutes, rules and regulations. In the foregoing sentence, the word "service' shall be construed to refer to the health care service rendered by the hospital or other health service facility.
- 28. (a) In accordance with the 1980 Omnibus Reconciliation Act (Public Law 96-499), Contractor hereby agrees that until the expiration of four years after the furnishing of services under this agreement, Contractor shall make available upon written request to the Secretary of Health and Human Services, or upon request, to the Comptroller General of the United States or any of their duly authorized representatives, copies of this contract, books, documents and records of the Contractor that are necessary to certify the nature and extent of the costs hereunder.
- (b) If Contractor carries out any of the duties of the contract hereunder, through a subcontract having a value or cost of \$10,000 or more over a twelve-month period, such subcontract shall contain a clause to the effect that, until the expiration of four years after the furnishing of such services pursuant to such subcontract, the subcontractor shall make available upon written request to the Secretary of Health and Human Services or upon request to the Comptroller General of the United States, or any of their duly authorized representatives, copies of the subcontract and books, documents and records of the subcontractor that are necessary to verify the nature and extent of the costs of such subcontract.
- (c) The provisions of this section shall apply only to such contracts as are within the definition established by the Health Care Financing Administration, as may be amended or modified from time to time.

Affirmative Action Clauses

exhibit a-1

State University of New York

1. **DEFINITIONS**. The following terms shall be defined in accordance with Section 310 of the Executive Law:

STATE CONTRACT herein referred to as "State Contract", shall mean: (a) a written agreement or purchase order instrument, providing for a total expenditure in excess of twenty-five thousand dollars (\$25,000.00), whereby the State University of New York ("University") is committed to expend or does expend funds in return for labor, services including but not limited to legal, financial and other professional services, supplies, equipment, materials or an combination of the foregoing, to be performed for, or rendered or furnished to the University; (b) a written agreement in excess of one hundred thousand dollars (\$100,000,00) whereby the University is committed to expend or does expend funds for the acquisition, construction, demolition, replacement, major repair or renovation of real property improvements thereon; and (c) a written agreement in excess of one hundred thousand dollars (\$100,000.00) whereby the University as an owner of a state assisted housing project is committed to expend or does expend funds for the acquisition, construction, demolition, replacement, major repair or renovation of real property and improvements thereon for such project.

SUBCONTRACT herein referred to as "Subcontract", shall mean any agreement providing for a total expenditure in excess of \$25.000 for construction, demolition. replacement, major repair, renovation. planning or design of real property and improvements thereon between a Contractor and any individual, partnership, corporation, or not-for-profit corporation, in which a portion of a Contractor's obligation under a State Contract is undertaken or assumed, but shall not include any construction, demolition, replacement, major repair, planning or design of real property and improvements thereon for the beneficial use of Contractor.

WOMEN-OWNED BUSINESS ENTER- PRISE

herein referred to as "WBE", shall mean a business enterprise, including proprietorship, partnership or corporation that is: (a) at least fifty-one percent owned by one or more United States citizens or permanent resident aliens who are women; (b) an enterprise in which the ownership interest of is real, substantial and women continuing: (c) an enterprise in which such women ownership has and exercises the authority to control independently the day-today business decisions of the enterprise; (d) an enterprise authorized to do business in state and independently owned and operated: (e) an enterprise owned by an individual or individuals, whose ownership, control and operation are relied upon for certification, with a personal net worth that does not exceed three million five hundred thousand dollars, as adjusted annually on the first of January for inflation according to the consumer price index of the previous year; and (f) an enterprise that is a small business pursuant to subdivision twenty of this section.

A firm owned by a minority group member who is also a woman may be certified as a minority-owned business enterprise, women-owned business enterprise, or both, and may be counted towards either a minority-owned business enterprise goal or a women-owned business enterprise goal, in regard to any Contract or any goal, set by an agency or authority, but such participation may not be counted towards both such goals. Such an enterprise's participation in a Contract may not be divided between the minorityowned business enterprise goal and the women-owned business enterprise goal.

MINORITY-OWNED BUSINESS ENTER-PRISE herein referred to as "MBE", shall mean a business enterprise, including a sole proprietorship, partnership or corporation that is: (a) at least fifty-one percent (51%) owned by one or more minority group members; (b) an enterprise in which such minority ownership real, substantial and continuing; (c) an enterprise in which such

June 6.

minority ownership has and exercises the authority to control independently the day-today business decisions of the enterprise; (d) an enterprise authorized to do business in this state and independently owned and operated; (e) an enterprise owned by an individual or individuals, whose ownership, control and operation are relied upon for certification, with a personal net worth that does not exceed three million five hundred thousand dollars. (\$3,500,000.00) as adjusted annually on the first of January for inflation according to the consumer price index of the previous year; and (f) an enterprise that is a small business pursuant to subdivision twenty of this section.

MINORITY GROUP MEMBER shall mean a United States citizen or permanent resident alien who is and can demonstrate membership in one of the following groups: (a) Black persons having origins in any of the Black African racial groups; (b) Hispanic persons of Mexican, Puerto Rican, Dominican, Cuban, Central or South American of either Indian or Hispanic origin, regardless of race; (c) Native American or Alaskan native persons having origins in any of the original peoples of North America. (d) Asian and Pacific Islander persons having origins in any of the Far East countries, South East Asia, the Indian Subcontinent or Pacific Islands.

CERTIFIED ENTERPRISE OR BUSINESS shall mean a business verified as a minority or women- owned business enterprise pursuant to section 314 of the Executive Law. A business enterprise which has been approved by the DMWBD for minority or women-owned enterprise status subsequent to verification that the business enterprise is owned, operated, and controlled by minority group members or women, and that also meets the financial requirements set forth in the regulations.

2. TERMS. The parties to the attached State Contract agree to be bound by the following provisions which are made a part hereof (the

Solar Purchase Power Agreement Template

word "Contractor" herein refers to any party other than the University:

- 1(a) Contractor and its Subcontractors shall undertake or continue existing programs of affirmative action to ensure that minority group members and women are afforded equal employment opportunities without discrimination. For these purposes, affirmative action shall apply in the areas of recruitment, employment, job assignment, promotion, upgrading, demotion, transfer, layoff, or termination and rates of pay or other forms of compensation.
- (b) Prior to the award of a State Contract, the Contractor shall submit an equal employment opportunity (EEO) policy statement to the University within the time frame established by the University.
- (c) As part of the Contractor's EEO policy statement, the Contractor, as a precondition to entering into a valid and binding State Contract, shall agree to the following in the performance of the State Contract: (i) The Contractor will not discriminate against any employee or applicant for employment, will undertake or continue existing programs of affirmative action to ensure that minority group members and women are afforded equal employment opportunities without discrimination, and shall make and document its conscientious and active efforts to employ and utilize minority group members and women in its work force on State Contracts;(ii) The Contractor shall state in all solicitations or advertisements for employees that, in the performance of the State Contract, all qualified applicants will be afforded equal employment opportunities without discrimination: (iii) At the request of the University the Contractor shall request each employment agency, labor union, or authorized representative of workers with which it has a collective bargaining or other agreement or understanding, to furnish a written statement that such employment agency, labor union, or representative will not discriminate, and that such union or representative will affirmatively cooperate in the implementation of the Contractor's obligations herein.
- (d) Except for construction Contracts, prior to an award of a State Contract, the Contractor shall submit to the contracting agency a staffing plan of the anticipated work force to be utilized on the State Contract or, where required, information on the Contractor's total work force, including apprentices, broken down by specified ethnic background, gender, and Federal occupational categories or other appropriate categories specified by the contracting agency. The form of the staffing plan shall be supplied by the contracting agency. If Contractor fails to provide a staffing plan, or in the alternative, a description of its entire work force, the University may reject Contractor's bid, unless Contractor either commits to provide such information at a later date or provides a reasonable justification in writing for its failure to provide the same.
- (e) After an award of a State Contract, the Contractor shall submit to the University a workforce utilization report, in a form and manner required by the agency, of the work force actually utilized on the State Contract, broken down by specified ethnic background, gender, and Federal occupational categories or

other appropriate categories specified by the University.

- The Contractor shall include the (f) provisions of this section in every Subcontract in such a manner that the requirements of the provisions will be binding upon each Subcontractor as to work in connection with the State Contract, including the requirement that Subcontractors shall undertake or continue existing programs of affirmative action to ensure that minority group members and women are afforded equal employment opportunities without discrimination, and, when requested, provide to the information on the ethnic background, gender, and Federal occupational categories of the employees to be utilized on the State Contract.
- To ensure compliance with the requirements of this paragraph, the University shall inquire of a Contractor whether the work force to be utilized in the performance of the State Contract can be separated out from the Contractor's and/or Subcontractors' total work force and where the work of the State Contract is to be performed. For Contractors who are unable to separate the portion of their work force which will be utilized for the performance of this State Contract. Contractor shall provide reports describing its entire work force by the specified ethnic background, gender, and Federal Occupational Categories, or other appropriate categories which the agency may specify.
- (h) The University may require the Contractor and any Subcontractor to submit compliance reports, pursuant to the regulations relating to their operations and implementation of their affirmative action or equal employment opportunity program in effect as of the date the State Contract is executed.
- (i) If a Contractor or Subcontractor does not have an existing affirmative action program, the University may provide to the Contractor or Subcontractor a model plan of an affirmative action program. Upon request, the Director of New York State Department of Economic Development, Division of Minority and Women Business Development (DMWBD) shall provide a contracting agency with a model plan of an affirmative action program.
- (j) Upon request, DMWBD shall provide the University with information on specific recruitment sources for minority group members and woman, and contracting agencies shall make such information available to Contractors
- 2. Contractor must provide the names, addresses and federal identification numbers of certified minority- and women-owned business enterprises which the Contractor intends to use to perform the State Contract and a description of the Contract scope of work which the Contractor intends to structure to increase the participation by Certified minorityand/or women-owned business enterprises on the State Contract, and the estimated or, if known, actual dollar amounts to be paid to and performance dates of each component of a State Contract which the Contractor intends to be performed by a certified minority- or woman-owned business enterprise. In the event the Contractor responding to University solicitation is joint venture, teaming agreement, or other similar arrangement that includes a minority-and women owned business enterprise, the Contractor must submit for

review and approval: i. the name, address, telephone number and federal identification of each partner or party to the agreement; ii. the federal identification number of the joint venture or entity established to respond to the solicitation, if applicable; iii. A copy of the joint venture, teaming or other similar arrangement which describes the percentage of interest owned by each party to the agreement and the value added by each party; iv. A copy of the mentor-protégé agreement between the parties, if applicable, and if not described in the joint venture, teaming agreement, or other similar arrangement.

- 3. PARTICIPATION BY MINORITY GROUP MEMBERS AND WOMEN. The University shall determine whether Contractor has made conscientious and active efforts to employ and utilize minority group members and women to perform this State Contract based upon an analysis of the following factors:
- (a) Whether Contractor established and maintained a current list of recruitment sources for minority group members and women, and whether Contractor provided written notification to such recruitment sources that contractor had employment opportunities at the time such opportunities became available.
- (b) Whether contractor sent letters to recruiting sources, labor unions, or authorized representatives of workers with which contractor has a collective bargaining or other agreement or understanding requesting assistance in locating minority group members and women for employment.
- (c) Whether contractor disseminated its EEO policy by including it in any advertising in the news media, and in particular, in minority and women news media.
- (d) Whether contractor has attempted to provide information concerning its EEO policy to Subcontractors with which it does business or had anticipated doing business.
- (e) Whether internal procedures exist for, at a minimum, annual dissemination of the EEO policy to employees, specifically to employees having any responsibility for hiring, assignment, layoff, termination, or other employment decisions. Such dissemination may occur through distribution of employee policy manuals and handbooks, annual reports, staff meetings and public postings.
- (f) Whether Contractor encourages and utilizes minority group members and women employees to assist in recruiting other employees.
- (g) Whether Contractor has apprentice training programs approved by the N.Y.S. Department of Labor which provides for training and hiring of minority group members and women.
- (h) Whether the terms of this section have been incorporated into each Subcontract which is entered into by the Contractor.
- 4. PARTICIPATION BY MINORITY AND WOMEN-OWNED BUSINESS ENTERPRISES. Based upon an analysis of the following factors, the University shall determine whether Contractor has made good faith efforts to provide for meaningful participation by minority-owned and womenowned business enterprises which have been certified by DMWBD:

- (a) Whether Contractor has actively solicited bids for Subcontracts from qualified M/WBEs, including those firms listed on the Directory of Certified Minority and Women-Owned Business Enterprises, and has documented its good faith efforts towards meeting minority and women owned business enterprise utilization plans by providing, copies of solicitations, copies of any advertisements for participation by certified minority- and women-owned business enterprises timely published in appropriate general circulation, trade and minority- or women-oriented publications, together with the listing(s) and date(s) of the publications of such advertisements; dates of attendance at any pre-bid, pre-award, or other meetings, if any, scheduled by the University, with certified minority- and women-owned business enterprises, and the reasons why any such firm was not selected to participate on the
- (b) Whether Contractor has attempted to make project plans and specifications available to firms who are not members of associations with plan rooms and reduce fees for firms who are disadvantaged.
- (c) Whether Contractor has utilized the services of organizations which provide technical assistance in connection with M/WBE participation.
- (d) Whether Contractor has structured its Subcontracts so that opportunities exist to complete smaller portions of work.
- e) Whether Contractor has encouraged the formation of joint ventures, partnerships, or other similar arrangements among Subcontractors.
- (f) Whether Contractor has requested the services of the Department of Economic Development (DED) to assist Subcontractors' efforts to satisfy bonding requirement.
- (g) Whether Contractor has made progress payments promptly to its Subcontractors.
- (h) Whether the terms of this section have been incorporated into each Subcontract which is entered into by the Contractor. It shall be the responsibility of Con-tractor to ensure compliance by every Subcontractor with these provisions.

5. GOALS. (a) GOALS FOR MINORITY AND WOMEN WORK FORCE PARTICIPATION.

- (i) The University shall include relevant work force availability data, which is provided by the DMWBD, in all documents which solicit bids for State Contracts and shall make efforts to assist Contractors in utilizing such data to determine expected levels of participation for minority group members and women on State Contracts.
- (ii) Contractor shall exert good faith efforts to achieve such goals for minority and women's participation. To successfully achieve such goals, the employment of minority group members and women by Contractor must be substantially uniform during the entire term of this State Contract. In addition, Contractor should not participate in the transfer of employees from one employer or project to another for the sole purpose of achieving goals for minority and women's participation.
- (b) GOALS FOR MINORITY AND WOMEN-OWNED BUSINESS ENTER-

PRISES PARTICIPATION. For all State Contracts in excess of \$25,000.00 whereby the University is committed to expend or does expend funds in return for labor, services including but not limited to legal, financial and other professional services, supplies, equipment, materials or an combination of the foregoing or all State Contracts in excess of \$100,000.00 whereby the University is committed to expend or does expend funds for the acquisition, construction, demolition, replacement, major repair or renovation of real property and improvements thereon, Contractor shall exert good faith efforts to _ percent achieve a participation goal of __%) for Certified Minority-Owned percent Business Enterprises and Certified Women-Owned _%) for Business Enterprises.

6. ENFORCEMENT. The University will be responsible for enforcement of Contractor's compliance with these provisions. Contractor, and each Subcontractor, shall permit the University access to its books, records and accounts for the purpose of investigating and determining whether Contractor or Subcontractor is in compliance with the requirements of Article 15-A of the Executive Law. If the University determines that a Contractor or Subcontractor may not be in compliance with these provisions, the University may make every reasonable effort to resolve the issue and assist the Contractor or Subcontractor in its efforts to comply with these provisions. If the University is unable to resolve the issue of noncompliance, the University may file a complaint with the DMWBD.

7. DAMAGES FOR NON COMPLIANCE.

Where SUNY determines that Contractor is not in compliance with the requirements of the Contract and Contractor refuses to comply with such requirements, or if Contractor is found to have willfully and intentionally failed to comply with the MWBE participation goals, Contractor shall be obligated to pay to SUNY liquidated damages. Such liquidated damages shall be calculated as an amount equaling the difference between:

- a. All sums identified for payment to MWBEs had the Contractor achieved the contractual MWBE goals; and
- b. All sums actually paid to MWBEs for work performed or materials supplied under the Contract.

In the event a determination has been made which requires the payment of liquidated damages and such identified sums have not been withheld by SUNY, Contractor shall pay such liquidated damages to SUNY within sixty (60) days after they are assessed by SUNY unless prior to the expiration of such sixtieth day, the Contractor has filed a complaint with the Director of the Division of Minority and Woman Business Development pursuant to Subdivision 8 of Section 313 of the Executive Law in which event the liquidated damages shall be payable if Director renders a decision in favor of SUNY.

Exhibit B

Purchaser acknowledges that Provider will be financing the installation of the System either through a lessor, lender or with financing accommodations from one or more financial institutions and that the Provider may sell or assign the System and/or may secure the Provider's obligations by, among other collateral, a pledge or collateral assignment of this Agreement and a first security interest in the System. Any such financing shall be conducted in accordance with the terms and conditions of the Agreement. In order to facilitate such necessary sale, conveyance, or financing, and with respect to any such financial institutions of which Provider has notified Purchaser in writing Purchaser agrees as follows:

- (a) <u>Consent to Collateral Assignment</u>. Purchaser consents to either the sale or conveyance to a lessor or the collateral assignment by Provider to the a lender that has provided financing of the System, of the Provider's right, title and interest in and to this Agreement, with such sale or conveyance to be conducted in accordance with Section 13.1 of the Agreement.
- (b) <u>Notices of Default</u>. Purchaser will deliver to the Financing Party, concurrently with delivery thereof to Provider, a copy of each notice of default given by Purchaser under the Agreement, inclusive of a reasonable description of Provider default.
- (c) <u>Rights Upon Event of Default</u>. Notwithstanding any contrary term of this Agreement:
 - (i) the Financing Party, as collateral assignee, shall be entitled to exercise, in the place and stead of Provider, any and all rights and remedies of Provider under the Agreement in accordance with the terms of the Agreement as if the Financing Party were the Provider and only in the event of a Provider Default. The Financing Party shall also be entitled to exercise all rights and remedies of secured parties generally with respect to the System;
 - (ii) the Financing Party shall have the right, but not the obligation, to pay all sums due under the Agreement and to perform any other act, duty or obligation required of Provider thereunder or cause to be cured any default of Provider thereunder in the time and manner provided by the terms of the Agreement as if such Financing Party were Provider; and
 - (iii) upon the exercise of remedies under its security interest in the System, including any sale thereof by the Financing Party, whether by judicial proceeding or under any power of sale contained therein, or any conveyance from Provider to the Financing Party (or any assignee of the Financing Party) in lieu thereof, the Financing Party shall give notice to Purchaser of the transferee or assignee of the Agreement, which shall be conducted in accordance with Section 13.1 of the Agreement as if the Financing Party were the Provider. Any such exercise of remedies shall not constitute a default under this Agreement.

(d) Right to Cure.

- (i) Purchaser will not exercise any right to terminate or suspend this Agreement unless it shall have given the Financing Party prior written notice by sending notice to the Financing Party (at the address provided by Provider) of its intent to terminate or suspend this Agreement, specifying the condition giving rise to such right. As provided for in Section (c) of this Exhibit B, the Financing Party shall have the right to cure any Provider Default within the time periods specified within the Agreement as if the Financing Party were the Provider.
- (ii) If the Financing Party (including any purchaser or transferee), pursuant to an exercise of remedies by the Financing Party, shall acquire title to or control of Provider's assets and shall, within the time periods described in Sub-section (c)(i) of this Exhibit B above, cure all defaults under the Agreement existing as of the date of such change in title or control in the manner required by this Agreement and which are capable of cure by a third person or entity, then such person or entity shall no longer be in default under the Agreement, and the Agreement shall continue in full force and effect.

EXHIBIT C

RFP

[Document follows this introductory page]

EXHIBIT D

Proposal

[Document to follow this introductory page.]

Schedule 1: Description of Premises and System

System Premises:

•

Anticipated Rebate or Subsidy:

- New York State Energy Research and Development Authority's most current solar PON
- Incentive Amount =

System Size:

Equipment:

- Racking:
- Modules:
- Inverter:
- Monitoring:

Scope:

Provider Responsibilities

- a) **Design Preparation:** Inspect the proposed System mounting site to assure long-term safety and stability.
- b) **Rebate:** Prepare all rebate documentation and communicate with all relevant agencies, if applicable.
- c) **Design:** Produce and provide CAD single-line electrical and layout drawings of the System. Drawings will consist of a complete site plan showing the location of the array, inverters, and routing of conduits, an elevation showing panel visibility from the street, and any details necessary for the plan check and permitting.
- d) **Pre-installation Conference:** Before System installation, conduct a pre-installation conference at the project site to review procedures, schedules, safety, and coordination of the installation. Several conferences may be needed if the complexities and construction schedules so require.
- e) **Procurement, Installation:** Furnish all necessary mounting hardware, photovoltaic modules, electrical equipment, and labor for installation of the System up to the utility grid interconnection point[s] as indicated in this Agreement.
- f) **Permits:** Obtain all permits required to perform the Work.
- g) **Inspections:** Serve as the owner's representative for applicable System inspections.

- h) **Site Safety:** Jobsite safety meetings once per week during Installation Work, or upon addition of new personnel to the jobsite.
- i) **Site Access:** During construction, Provider will be responsible for the security of the system, equipment, tools and other materials related to the Provider's Work.

j) Acceptance Testing and System Commissioning: Provider shall:

- i. Conduct an inspection, test and commissioning procedure to insure that the System is installed in a professional manner and consistent with prudent industry practices. A record of the installation and the major components including modules, inverters, transformers, and source circuit combiners will be documented in a test and commissioning report.
- ii. Test and verify that all non-current-carrying metal parts are solidly grounded and all equipment and System grounding is installed and functional per NEC 2008.
- iii. Test and verify that phase sequencing, fuse continuity, and open circuit voltage are within the manufacturers recommended range at the DC disconnect.
- iv. Test and verify that the inverter is operating effectively within the typical start up time and record the DC operating voltage, phase currents, and inverter power.
- v. Provide complete operation and maintenance manual for the System (two printed copies and one electronic copy). The manual will include: (i) as-built drawings, (ii) as-built shop drawings, (iii) a copy of any required submittals or filing, (iv) product cut sheets, (v) product operation manuals, (vi) a copy of the photo record, (vii) written Utility approval, (viii) product warranties; and (ix) supplier and installer contact information.
- k) Solar Monitoring System: Every COMPANY System comes standard with MONITORING DEVICE proprietary solar monitoring service. MONITORING DEVICE enables COMPANY to continuously monitor the key performance variables of the System and transmit this data to COMPANY servers through the Internet. All Purchaser systems will be accessible through a single data access including system-level and aggregated system data. Any additional third-party monitoring, network access or data feed will be cost and responsibility of Purchaser.

2. Purchaser Responsibilities

- a) **Site Preparation:** Responsible for pre-installation site clean-up including removal of all debris and obstacles that may impede System installation.
- b) **Site Access:** Responsible for security access to the System after installation. During construction, Provider will be responsible for the security of the system, equipment, tools and other materials related to the Provider's Work.
- c) **Existing Facilities:** Provide all available design and as-built drawings of the existing facilities to facilitate System design and installation.
- d) **System Management:** Provide a single point of contact for each installation. Purchaser point of contact shall have authority for all written requests for project changes and shall be available on 24 hours' notice. In the event that Provider is providing its services as a subcontractor and its work is part of a larger program, Purchaser will keep Provider

informed about the overall progress of the program for which Provider's work is a part, and coordinate Provider's work with the work being performed by Purchaser's other contractors, including giving Provider sufficient advance notice (at least 5 business days) of when Provider is to perform the work under this Agreement. The foregoing shall not limit Purchaser's obligations under the terms and conditions of this agreement.

3. Clarifications

- a) **Work Hours:** Provider's standard work hours are Monday through Friday 7am to 5:30pm unless approved by Provider's System Manager.
- b) **Included in pricing:** COMPANY Performance Guarantee, Installation of a solar energy system (includes: design, engineering, permitting, installation, monitoring, rebate application and paperwork processing for solar energy system). Payment of prevailing wage labor.
- c) **Exclude in pricing:** Unforeseen groundwork (including, but not limited to, excavation/circumvention of underground obstacles), upgrades or repair to Purchaser or utility electrical infrastructure, payment bonds, performance bonds, tree removal, tree trimming, the payment of special labor wages (i.e. union wages not required).

Schedule 2 - - kWh Rate

The kWh Rate with respect to the System under the Agreement shall be in accordance with the following schedule:

Year of System	kWh Rate
Term	(\$/kWh)
1	
2	
3	
4	
5	
6	
7	
8	
9	
10	
11	
12	
13	
14	
15	
16	
17	
18	
19	
20	
21	
22	
23	
24	
25	

Schedule 3 – Early Termination Fee & Option Price

The Early Termination Fee and Option Price with respect to the System under the Agreement shall be calculated in accordance with the following:

Column 1 Early Termination Fee	
Contract Year	Early Termination Value
1	
2	
3	
4	
5	
6	
7	
8	
9	
10	
11	
12	
13	
14	
15	
16	
17	
18	
19	
20	
21	
22	
23	
24	
25	

Column 2 Option Price*		
End of Contract Year	Option Price*:	
6		
10		
15		
20		
25	Fair Market Value	

*In accordance with Section 2.3 of the Agreement, the Option Price shall be the greater of the amount set forth above and the Fair Market Value of the System as determined pursuant to Section 2.3

^{*}Includes Early Termination prior to the Commercial Operation Date.

Schedule 4 – Estimated Annual Production

Estimated Annual Production commencing on the Commercial Operation Date with respect to System under the Agreement shall be as follows:

Year of System	
Term	Estimated Production (kWh)
1	
2	
3	
4	
5	
6	
7	
8	
9	
10	
11	
12	
13	
14	
15	
16	
17	
18	
19	
20	
21	
22	
23	
24	
25	

The values set forth in this table are estimates (and not guarantees), of approximately how many kWhs are expected to be generated annually by the System.

Schedule 5 – Identification of Provider's Financing Party

Provider shall provide Purchaser with notice information for any Financing Party within a reasonable time after such Financing Party assumes any interest in the System.

Schedule 6 – Notice Information

Purchaser:

SUNY Purchase

Nikolaus Lentner, Purchasing Manager 735 Anderson Hill Rd Purchase, NY 10577

Phone: 914-251-6070 Fax: 914-251-6076

Email: Nikolaus.lentner@purchase.edu

With a copy to:

Provider:

COMPANY ADDRESS CITY STATE ZIP CODE PHONE FAX EMAIL

With a copy to:

[]

Financing Party:

[To be provided by Provider if applicable and when known]